Cooperative Formation Control of Autonomous Underwater Vehicles: An Overview

Bikramaditya Das¹ Bidyadhar Subudhi² Bibhuti Bhusan Pati¹

¹Department of Electrical Engineering, Veer Surendra Sai University of Technology, Burla 768018, India ²Department of Electrical Engineering, National Institute of Technology, Rourkela 769008, India

Abstract: Formation control is a cooperative control concept in which multiple autonomous underwater mobile robots are deployed for a group motion and/or control mission. This paper presents a brief review on various cooperative search and formation control strategies for multiple autonomous underwater vehicles (AUV) based on literature reported till date. Various cooperative and formation control schemes for collecting huge amount of data based on formation regulation control and formation tracking control are discussed. To address the challenge of detecting AUV failure in the fleet, communication issues, collision and obstacle avoidance are also taken into attention. Stability analysis of the feasible formation is also presented. This paper may be intended to serve as a convenient reference for the further research on formation control of multiple underwater mobile robots.

Keywords: Autonomous underwater vehicles (AUV), cooperative control, formation control, tracking control, regulatory control.

1 Introduction

An autonomous underwater vehicles (AUV) is a submersible underwater vehicle which is equipped with power supply and is controlled by an onboard computer while performing a task. AUVs are compact, independent and low-drag profile crafts powered by underwater direct current(DC) power thrust^[1]. AUVs are equipped with sensors to measure temperature, salinity, and pollutant concentration. The underwater vector magnetic fields produced by AUV are as shown in Fig. 1. Motion control of AUVs can be of different types such as based on navigation, path tracking and formation^[2]. AUVs are powered by rechargeable batteries (e.g., lithium polymer, nickel metal hydride, lithium ion, aluminum based semi fuel cells, etc.), depending on the length of motion plan. To collect data for mission, various sensors are used for example compasses, depth sensor, side scan sonar, magnetometers, accelerometers, gyro meters, thermistors, doppler velocity log (DVL) and conductivity $probes^{[3]}$.

A number of autonomous underwater vehicles are deployed for a group motion and/or control mission for several important activities such as pipeline inspections in gas and oil industries, oceanographic observations, bathymetric surveys, military usages, recovery of lost man-made objects, high resolution seabed inspection, mapping, commercial survey and neutralization of undersea mines area. To accomplish the cooperative motion successfully, formation control is considered as an important cooperative control paradigm^[4].



Fig. 1 Simplified block diagram of an AUV architecture^[4]

The word autonomous refers to execution of the assigned mission without any external human intercession. AUVs are also known as unmanned undersea vehicles (UUVs) and remotely operated underwater vehicles (ROVs) depending upon their control for the assignment. This means that UUVs are autonomous, whereas non-autonomous remotely operated underwater vehicles controlled and powered from the surface by an operator/pilot are known as ROVs. Based on control structure, AUVs are of two types, fully actuated system and under actuated system^[5]. Control laws for a fully-actuated system are developed by using the control allocation map. In an under actuated system, it is challenging to develop a control law together with ensuring the system stability. For both the cases, it is necessary to show the robustness and adaptation of the control structure for the external disturbances. Formation control is an emerging research topic in robotics focusing on controlling the relative positions, velocity and orientations of AUVs to move in a group. It is aimed to accomplish tasks such as mapping, exploration, monitoring of marine environments, data collection for oceanographic missions, security patrols and autonomous navigation information^[6]. It increases the

Review Manuscript received August 6, 2015; accepted December 18, 2015. Recommended by Associate Editor Veljko Potkonjak

 $[\]bigodot$ Institute of Automation, Chinese Academy of Sciences and Springer-Verlag Berlin Heidelberg 2016

robustness and efficiency to achieve the task by enhancing the reconfiguration ability and structure flexibility. In military missions, a group of autonomous underwater vehicles are required to keep in a specified formation for area coverage and exploration^[7]. In many applications, a group of AUVs are intended to follow a predefined trajectory while maintaining a desired spatial pattern. Due to uncertainty in the AUV dynamics for wave disturbances and communication constraints in the acoustic medium, it is difficult to control AUVs^[8, 9].

The idea of formation control comes from nature such as swarming behavior of living beings, flocks of birds, school of fishes, herds of wild beasts and colonies of bacteria^[10]. Observing the formation, it is found that when they move as a team, they must avoid colliding with each other and with a common average heading. Each AUV has a local control strategy, to steer towards the average heading of neighbors, steer to avoid crowding and steer towards the average position of neighbor to achieve alignment, separation and cohesion^[11]. Underwater mobile robots were used for data gathering and transferring in early 1960. Then around 1970, the first AUV named as Torpedo was developed and used only for test purposes in USA^[12]. In mid 1980s, implementation of theoretical developments on AUV were made in the practical fields. Commercialization and development of cooperative motion control of multiple AUVs in both theoretical and practical development are achieved from 1980 till this date as shown in Fig. $2^{[13]}$.



Fig. 2 Studies on formation control of AUVs

A formation control problem considers the following typical aspects: assignment of feasible formation, maintenance of formation shape and switching between formations. Formation control may be classified as formation regulation control and formation tracking control^[14]. Formation of a group of AUVs is called a rigid formation where formation structure remains fixed or flexible due to presence of obstacles throughout moving period, which is known as formation regulation control^[15]. In order to maintain a group of AUVs in a formation following the desired trajectories, each AUV needs to communicate with its neighbors. This type of formation is known as formation tracking $control^{[16-18]}$. To achieve a successful formation control of multiple AUVs, the following steps should be followed such as choosing of appropriate AUVs models (both kinematics and dynamics) to stay in formation group, regulatory/trajectory coordination strategies^[19-21]</sup>. An AUV and/or a group of AUVs in formation or cooperation, play an important role for different applications such as security patrols, search and rescue in hazardous environments^[14,17,20]. In military missions, a group of autonomous vehicles is required to keep a specified formation for area coverage and reconnaissance. In small clustering, formation helps to reduce the fuel consumption for propulsion and expand their sensing capabilities. The path following is the method of forcing an AUV along the desired or predefined path without any time constraint. By this method, the AUV starts from an initial point with a definite orientation to reach at the destination point. Sometimes, the controlling of orientation may be avoided according to the necessary degrees of freedom and the directions are controlled by the planner orientations only^[16]. In the case of the position controlling problems, the AUVs are directed to merge with the desired positions of the desired path. But in the speed controlling problems, the forward speeds are forced to the desired speeds.

To achieve a successful cooperative control of multiple AUVs, the prerequisite points are chosen by the AUVs to stay in formation group, trajectory^[17]. Based on the above, the cooperative control is of two types: 1) formation control of multiple AUVs and 2) flocking control of multiple AUVs. Formation control is an important research topic of the cooperative control within the recent fields of multi-AUV systems^[14]. The formation control is referred as the problems of controlling the relative positions and orientations of AUVs in a group while allowing the group to move as a whole as shown in Fig. 3. Formation control consists of the following steps: 1) assignment of feasible formation, 2) maintenance of formation shape to move in formation, 3) switching between formations.



Fig. 3 Formation control of multiple AUVs

Flocking is the flying behavior of a flock of birds, which is a collective behavior of living beings such as schooling of fishes, flocks of birds, grouping of insects, colonies of bacteria and herds of animals, etc. This can be applicable to design control algorithm for a group of multiple AUVs to perform a desired task as shown in Fig. 4. Flocking control of multiple AUVs is similar to that of formation control with the only difference is that there are no constraints on distance among AUVs^[18–20]. In case of formation control, the distances among AUVs are always fixed. Our research work only focuses on formation control of multiple AUVs^[21].



Fig. 4 Flocking control of multiple AUVs

Cooperative control may be classified as formation control and flocking control. The path following is the method of forcing the AUV along the desired predefined path without any time constraint. Flocking control may be achieved by a group of multiple AUVs to perform a desired task. Formation control has broad applications and so it is recognized to occupy the seat of an active research topic in the recent years. The steps for achieving formation control are: 1) assignment of feasible formation, 2) moving in formation, 3) maintenance of formation shape, 4) switching between formations. The definition of formation is specified in three different ways. The ways may be meant to achieve the rigid formation [10], or to achieve a flexible formation in plane^[22]. In formation control, it is assumed that information is available for each AUV through local sensors within the formation. In the case of trajectory tracking problems, the follower AUVs are forced to merge a time parameterized geometric path. The control law is designed according to the temporal variations of the desired geometric path^[22]. The degree of difficulty of designing control law for formation of underactuated vehicles is very high. The trajectory tracking approach provides better performance for carrying control information on time as compared to the other path formation problem. In the virtual structure approach, the entire formation is treated as a single entity. Control methods are developed to force a group of AUVs to behave in a rigid formation. In virtual structure approach, the control law is derived in three $steps^{[23]}$. Also the convergence of the AUV to be in formation takes less time than that of the trajectory tracking problem.

The basic principle of behavior based approach is that, each AUV consists of a basic structure called motor schemas^[24]. The motor schemas generate their corresponding desired behavior. Some of the motor schemas are collision avoidance, formation shape and goal seeking. In formation control, the behavior-based approach and potential field approach are considered for various applications^[25-27]. The control input for the AUV is generated by adjusting using optimization technique. In [23], the virtual structure method is combined with the leader following method and behavioral approach to formation control. In this method, independent paths are derived from the desired path for each AUV^[28]. The methodology behind this method requires that the coordination must be achieved for AUVs following their respective desired paths^[29, 30]. In other words, it can be said that there exists a velocity and acceleration constraint for each derived path^[31].

Only a few papers are available in this field that includes discussions on formation control until 2014. Depending upon the above criteria, the contributions of this paper are as follows. Extensive reviews on formation control of multiple AUVs are listed in this paper. The objectives of this paper are as follows.

1) A detailed description about the system model of AUV in six degree of freedom (DOF) of earth fixed frame is discussed.

2) A peer review on formation control of multiple AUVs based on regulatory and tracking control techniques are briefly explained and stability analysis is also presented^[32-37].

3) Various challenges and applications of formation control are also discussed.

This paper is organized as follows. Section 2 describes the kinematics and dynamics of the AUV. Section 3 gives the problem formulation. A detailed description about coordination strategies for formation control is described in Section 4. Section 5 explains various challenges and formation failure issues. Section 6 describes the stability analysis of formation control. Section 7 discusses the various applications of formation control. Section 8 presents the conclusion of this paper.

2 AUV dynamics

Consider the schematic diagram of an AUV as shown in Fig. 5. The mathematical model (nonlinear equation of motion) of identical AUVs under the influence of wave



Fig. 5 Inertial and body fixed frames of an AUV model^[1]

disturbances in six degrees of freedom can be described as [1, 18-20].

$$\dot{\eta}_{1i} = J_1(\eta_{2i})\nu_{1i} \tag{1}$$

$$\dot{\eta}_{2i} = J_1(\eta_{2i})\nu_{2i}.$$
(2)

For multiple AUVs, $i = 1, 2, 3, \dots, N$, $\eta_i = [\eta_{1i}, \eta_{2i}]^{\mathrm{T}}$ is the position and orientation vector of *i*-th AUV in inertial frame, $\eta_{1i} = [x_i, y_i, z_i]^{\mathrm{T}}$ are the coordinates of positions and $\eta_{2i} = [\varphi_i, \theta_i, \psi_i]^{\mathrm{T}}$ is orientation along longitudinal, transversal, and vertical axes, respectively. $v_i = [v_{1i}, v_{2i}]^{\mathrm{T}}$ is the velocity vector with coordinates in the body-fixed frame, $v_{1i} = [u_i, v_i, w_i]^{\mathrm{T}}$ denotes linear velocities, $v_{2i} = [p_i, q_i, r_i]^{\mathrm{T}}$ are the angular velocities^[1].

$$\dot{\nu}_{1i} = M_1^{-1} (-C_{1i}(\nu_{1i})v_{2i} - D_{1i}(\nu_{1i})v_{1i} - g_1(\eta_{2i}) + \tau_{1i})$$

$$(3)$$

$$\dot{\nu}_{2i} = M_2^{-1} (-C_{1i}(\nu_{1i})v_{1i} - C_{2i}(\nu_{2i})v_{2i} - D_{2i}(\nu_{2i})v_{2i} - g_2(\eta_{2i}) + \tau_{2i})$$

$$(4)$$

 $\tau_i = [\tau_{1i}, \tau_{2i}]^{\mathrm{T}}$ is the control input to the *i*-th AUV, $\tau_{1i} = [X_i, Y_i, Z_i]^{\mathrm{T}}$ represents the external forces and $\tau_{2i} = [K_i, M_i, N_i]^{\mathrm{T}}$ denotes moments of external forces acting on the AUV for translational and rotational motion respectively. $J(\eta)$ is the non-singular transformation matrix, used for transformation from body fixed frame to earth fixed frame^[1]. The inertia mass matrices of AUV are given by

$$M_1 = \operatorname{diag}(m_{11}, m_{22}, m_{33}) \tag{5}$$

$$M_2 = \text{diag}(m_{44}, m_{55}, m_{66}). \tag{6}$$

The Coriolis and centripetal matrices are given by^[1]

$$C_{1i}(\nu_{1i}) = \begin{bmatrix} 0 & m_{33}w & -m_{22}v \\ -m_{33}w & 0 & m_{11}u \\ m_{22}v & -m_{11}u & 0 \end{bmatrix}$$
(7)

$$C_{2i}(\nu_{2i}) = \begin{bmatrix} 0 & m_{66}r & -m_{55}q \\ -m_{66}r & 0 & m_{44}p \\ m_{55}q & -m_{44}p & 0 \end{bmatrix}.$$
 (8)

The damping matrices including the added mass matrix $\mathrm{are}^{[1]}$

$$D_{1}(\nu_{1}) = \operatorname{diag}(d_{11} + \sum_{i=2}^{3} d_{ui} |u|^{i-1}, d_{22} + \sum_{i=2}^{3} d_{vi} |v|^{i-1}, d_{33} + \sum_{i=2}^{3} d_{wi} |w|^{i-1})$$
(9)

$$D_{2}(\nu_{2}) = \operatorname{diag}(d_{44} + \sum_{i=2}^{3} d_{pi} |p|^{i-1}, d_{55} + \sum_{i=2}^{3} d_{qi} |q|^{i-1}, d_{66} + \sum_{i=2}^{3} d_{ri} |r|^{i-1}).$$
(10)

Deringer

The restoring force and moment vector are given by $^{[1]}$

$$g_{2}(\eta_{2i}) = \begin{bmatrix} (W-B)s\theta, -(W-B)c\theta s\varphi, -(W-B)c\theta c\varphi \end{bmatrix}^{\mathrm{T}}$$

$$(11)$$

$$g_{2}(\eta_{2i}) = \begin{bmatrix} -(y_{G}W-y_{B}B)c\theta c\varphi + (z_{G}W-z_{B}B)c\theta s\varphi \\ (z_{G}W-z_{B}B)s\theta + (x_{G}W-x_{B}B)c\theta c\varphi \\ -(x_{G}W-x_{B}B)c\theta s\varphi - (y_{G}W-y_{B}B)s\theta \end{bmatrix}^{\mathrm{T}}$$

$$(12)$$

where W is the submerged weight and B is the buoyancy force of each identical AUV^[1]. $r_G = [x_G, y_G, z_G]^T$ is the distance between center of gravity of AUV and the origin of the body fixed frame and $r_B = [x_B, y_B, z_B]^T$ is the distance between center of buoyancy of AUV and the origin of the body fixed frame. The rotation matrix of the AUV can be expressed as^[1]

$$J_{1}(\eta_{2i}) = \begin{bmatrix} c\psi_{i}c\theta_{i} & -s\psi_{i}c\theta_{i} + s\phi_{i}s\theta_{i}c\psi_{i} & s\psi_{i}s\phi_{i} + s\theta_{i}c\psi_{i}c\phi_{i} \\ s\psi_{i}c\theta_{i} & c\psi_{i}c\phi_{i} + s\phi_{i}s\theta_{i}s\psi_{i} & -c\psi_{i}s\phi_{i} + s\theta_{i}s\psi_{i}c\phi_{i} \\ -s\theta_{i} & s\phi_{i}c\theta_{i} & c\phi_{i}c\theta_{i} \end{bmatrix}$$
(13)
$$J_{2}(\eta_{2i}) = \begin{bmatrix} 1 & s\phi_{i}t\theta_{i} & c\phi_{i}t\theta_{i} \\ 0 & c\phi_{i} & -s\phi_{i} \\ 0 & s\phi_{i}/c\theta_{i} & c\phi_{i}/c\theta_{i} \end{bmatrix} .$$
(14)

3 Problem statement

Control coordination issue relates to the study of the position, path planning and coordination within a defined communication topology in space and time respectively. It is assumed that the common velocity and position signals must be available to all the AUVs with the help of high level sensors. In addition to the above, underwater acoustic channels are affected by the long propagation delay of the acoustic signal, path loss, noise, multipath fading, Doppler spread, and high error probability. Various control



Fig.6 Control signal flow for formation control of leader-follower AUV $\,$

strategies are discussed which enable the follower AUVs to follow the leader as shown in Fig. 6. The formation control referred as the problems of controlling the relative positions and orientations of robots in a group while allowing the group to move as a whole. The information of the desired path of the leader AUV should be known to the follower AUVs to follow the leader with respect to position and orientation. In unreliable underwater network scenarios, it is very challenging to provide full observation communication protocol with energy-efficient reliable data transfer for time critical applications such as coordination. The factors affecting formation control are: 1) assignment of feasible formation control strategies, 2) moving in formation in presence of obstacle rich environment, 3) maintenance of formation shape availing full observation communication, 4) switching between formations.

4 Classification of formation control of AUVs

Generally, formation of AUVs may be categorized as path planning, path following and path tracking^[7, 19, 38]. The formation control of AUVs may be static or dynamic depending upon some regulation, whereas, reaching at the target point, the formation structure travels through a particular path known as formation tracking control. According to the above classification, some formation control techniques are enumerated till date as shown in the Fig. 7.



Fig. 7 Classification of formation control of multiple AUVs

The following techniques are some of the widely used formation control applied on underwater systems such as mapping, exploration, monitoring of marine environments, data collection for oceanographic missions and harbour security, $etc^{[39, 40]}$. For both the cases, such as formation regulation control and tracking control, it is necessary to show the robustness and adaptation of the control structure as shown in Fig. 8.



Fig. 8 Studies on formation control of AUVs

4.1 Formation regulation control

A specific shape needs to be maintained in a formation^[39]. It is essential to know the control issues based on regulatory control when multiple AUVs are in a cooperative motion. The different techniques for control regulation are presented below and a comparison study is made on various regulatory based formation control strategies as provided in Table 1.

4.1.1 Gateway coordination

In this configuration, an AUV serves as a gateway. Each of the AUV sends its information to gateway and receives the gateway's position. Coordination is thus achieved through the gateway AUV as shown in Fig. 9^[40]. The numbers represent the name of the AUVs, and the arrows present the coordination between the AUVs with their references.

4.1.2 Control abstraction based on neighbor's reference

An AUV needs to keep a proper position with respect to the position of neighbor AUVs to maintain formation. In Fig. 10, the coordination between the AUVs is shown with their neighbor's reference such as predecessor, leader and neighbor reference types^[13].

According to Wang et al.^[40], each AUV communicates to its immediate neighbors for coordination^[40] as shown in Fig. 11.



Fig. 9 Gateway coordination of multiple $\mathrm{AUVs}^{[13]}$



Fig. 10 Different neighbor reference types^[40]



Fig. 11 Spatial pattern of a group of three AUVs, an AUV with the preceding and succeeding AUVs

The n AUVs are indexed according to the spatial pattern they are required to achieve and communicate with the preceding and succeeding AUVs. Different control laws have been reported for the formation control of AUVs such as

1) cooperative control laws applicable to under actuated AUVs with a stationary formation is presented in [41]. 2) Robustness of the cooperative control laws is discussed with communication delays in [41].

Along with the group of AUVs coming into the stationary geometric pattern, each AUV is also required to converge to the same constant orientation as shown in Figs. 12 and 13.

4.1.3 Centralized approach

Here we consider the formation control of multiple AUVs according to a stationary geometric pattern under different communication scenarios. Centralized cooperative control laws are proposed with the aid of suitable transformations and the results on the graph theory as shown in Fig. $14^{[39]}$. An efficient formation control for the cooperative motion of

B. Das et al. / Cooperative Formation Control of Autonomous Underwater Vehicles: An Overview



Fig. 12 Spatial pattern of three AUVs using communication digraph



Fig. 13 Spatial pattern of AUVs with virtual agent using communication digraph



Fig. 14 Coordination of AUVs through a centralized approach $^{\left[39\right] }$

AUVs based on kinematic algorithm for the joint motion of an AUV with a leader AUV has been proposed in [42].

4.1.4 Neural network (NN) based dynamic surface control

Several formation control algorithms rely on the velocity and moment information of leader AUV. However, all AUVs are not equipped with velocity sensors. A NN based formation controller using back propagation learning algorithm eliminates the tracking errors of AUVs whose dynamics are highly nonlinear and time varying^[43]. It has good capability to incorporate the dynamics of the system. The proposed NN architectures have been designed to control the test bed for AUV named as naval postgraduate school(NPS) AUV.

4.1.5 Followers within a cluster

A leader-follower based on real-time communication design is adopted for the formation controller. Cluster space state method may be used for achieving formation control of underwater mobile robots as shown in Fig. $15^{[44]}$.



Fig. 15 Cluster based formation control, (a) linear, (b) triangular, (c) inverted triangle^[44]

In this method, the whole team of AUVs is divided into different clusters of limited sizes. A local controller is first designed for each cluster, and then formation control is developed by using centralized method^[45]. The implementation process of the leader-follower formation control algorithms are discussed such as line-shape, triangle-shape and inverted triangle-shape.

4.1.6 Spatially synchronized parallel formation

A synchronized parallel formation for a fleet of AUVs is developed in [46] which are based on its velocity matching and virtual leader control design. The spatial synchronized parallel formation is the basis of AUVs group's cooperative control, and is important for group cooperative behaviors, such as formation keeping in the space, communication distance fixing, and observation synchronization as shown in Fig. $16^{[47]}$.



Fig. 16 Spatially synchronized parallel formation

4.1.7 Surveillance on a path using non-hierarchical control system

In the formation control literature, two different types of formation control structures are used based on distribution of control tasks among agents. These are hierarchical and non-hierarchical structures^[48]. For a non-hierarchical structure, the distance maintaining and path tracking tasks are uniformly distributed across the formation, i.e., the control tasks distributed to the agents are relatively identical.

Stable periodic formation control of multiple nonholonomic vehicles may be achieved by arranging the vehicles in a cyclic interconnection topology known as "cyclic pursuit"^[49]. This cyclic pursuit could be a circular formation either with a fixed^[50] or varying radius^[51].

4.1.8 Stationary surveillance with formation change

In this formation, the shape maintained by the group of AUVs is rigid throughout. It is necessary to change or to split the shape of the formation if any static or dynamic obstacles come across. Static and dynamic obstacles are given by strong currents, land areas or heavy traffic shipping routes. For example, formation shape may be switched from straight line to wedge shape for stationary surveillance^[52] or may be changed from double platoon to other shape^[53]. By using the phase waves and phase gradient method the shape of the formation structure can be adapted in a similar fashion as an amoeba in drastic environmental condition. The shape of the formation can be changed and adapted to avoid obstacles by manipulating the potential functions associated with this as shown in Fig. $17^{[54]}$.



Fig. 17 A group of AUVs change their formation shape to pass through a narrow region

4.1.9 Cross-track formation control

The sliding mode controller proposed by Defoort et al. enables the AUVs to track the desired path at constant speed^[55]. Steps followed in [55] to track the desired path are as follows:

1) The first step is based on a line of sight guidance law as in [56, 57], which makes every AUV asymptotically follow a straight line path corresponding to the desired formation motion as shown in Fig. 18. Line of sight range can be estimated by visual sensors. In this range, each AUV can measure its speed and heading angle with respect to other AUVs.



Fig. 18 Line-of-sight and collision-avoidance path components

2) In the second step, the forward speed of every AUV is manipulated in such a way that they asymptotically converge to the desired formation and move with a desired forward speed profile as shown in Fig. 19.



Fig. 19 Group of AUVs change their formation shape to pass through a narrow region

4.1.10 Geometric formation control for AUVs

A group of AUVs may be allocated to reach at the destination through geometrical formation without any collision^[58-65]. Geometric formation control of multiple AUVs may be possible by utilizing the Jacobi shape theory as shown in Fig. 20^[19, 49, 66], velocity optimization technique^[67, 68] and path following control of single AUV are presented in [8, 69-72]. To solve the coordinated path following problem, a hybrid controller is to be developed as shown in Fig. 21^[73, 74].



Fig. 20 Jacobi vectors for three AUVs

B. Das et al. / Cooperative Formation Control of Autonomous Underwater Vehicles: An Overview



Fig. 21 Six AUVs moving in hexagonal geometrical formation^[46]

4.1.11 A region/boundary-based geometric formation control (or asset protection) scheme

The boundary based geometric path following problem of multiple AUVs can be developed for inspection as in [73–75]. In [76], a simple control approach based on a region boundary technique for geometric formation of multiple AUVs is presented.

The control objective is to keep each underwater vehicle at each corner of a desired geometric shape, i.e., an equilateral triangle or a square. An edge-based segmentation approach is utilized rather than specifying the minimum distance between members to ensure that each AUV is placed exactly at the desired position in their formation^[63, 77]. This allows each vehicle that has its own function to carry out an effective individual task^[78, 79], thus improving the formation performance as shown in Fig. 22.



Fig. 22 An illustration of task for AUVs

4.1.12 Behavioural approach of formation

The behavioral approach starts from behavior of individual AUV as shown in Fig. $23^{[80]}$. The common behaviors are goal seeking, obstacle avoidance, keeping the consistent formation, etc.

Therefore, more complex motion pattern can be generated by using the individual behavior of separate AUVs. The architecture of behavior based formation control of a team of AUVs consists of three levels, i.e., team behavior or team formation pattern, AUV task and behavior, and AUV $control^{[81, 82]}$.



Fig. 23 Control structure of formation control of multiple vehicles via behavior-based approach and potential field approach^[80]

4.2 Formation tracking control

4.2.1 Leader-follower approach

In this formation tracking control technique, an AUV is elected as the leader AUV, executes a path following algorithm at a required forward speed and relays its position to the remaining $AUVs^{[40, 83]}$. It is up to the followers to keep the formation, based on information received from the leader as shown in Fig. 24.



Fig. 24 An illustration of task for AUVs

4.2.2 Virtual leader-follower hierarchy structure

The dynamic behavior of leader represents the whole cluster behavior. If the leader behavior within a certain period is predefined, other AUVs within the cluster just obey the leader to track the leader's trajectory as shown in Fig. $25^{[84, 85]}$.

4.2.3 Formation coordinated control in the presence of communication losses

This paper addresses the problem of steering a group of AUVs along the given paths while holding a desired intervehicle formation pattern, all in the presence of communication losses as shown in Fig. $26^{[67]}$. The dynamics of each autonomous underwater vehicle can be dealt with each AUV controller locally. Coordination can then be achieved by resorting to a decentralized control law whereby the exchange of data among the vehicles is kept at a minimum.

🖄 Springer



Fig. 25 An illustration of task for AUVs



Fig. 26 Path following of *i*-th AUV with communication losses

4.2.4 Fuzzy logic based behavior fusion for multi-AUV formation keeping in uncertain ocean environment

This section presents a new behavior fusion method using fuzzy logic for coordinating multiple reactive behaviors as shown in Fig. $27^{[27]}$. The inputs to the proposed fuzzy control scheme for the leader AUV in multi-AUV system consist of the deviation in yaw angle while performing obstacle avoidance and goal seeking action separately, and the fuzzy control scheme for the follower AUV consists of the deviations in yaw angle while performing obstacle avoidance and formation keeping action separately^[86].

4.2.5 Synchronized path following control for multiple under actuated AUVs

The synchronized path following based control laws are categorized into two envelopes such as: 1) steering individual AUV to trace along predefined paths, 2) ensuring tracked paths of multiple AUVs to be synchronized, by



(c) Navigation supervisor for followers

Fig. 27 Fuzzy logic based multi-AUV formation^[27]

means of decentralized speed adaption under the constraints of multi-AUV communication topology as shown in Fig. 28^[87, 88]. With these two tasks formulation, geometric paths following are built on Lyapunov theory and back-stepping techniques for a class of individual path following control^[89–91]. Synchronization of path parameters are obtained by using a mixture of tools from linear algebra, graph theory and control theory^[8, 77, 92].

4.2.6 Decentralized overlapping tracking control of a formation of AUVs

This method is a new methodology based on the expansion and contraction paradigm as presented. The methodology is based on a specific linear formation model. Decentralized controllers for the extracted subsystems are contracted to the original position accordingly as shown in Fig. 29. The dynamic output feedback control law based on decentralized observers may also be used as shown in Fig. $30^{[93-95]}$.



Fig. 28 Synchronized path following control



Fig. 29 Decentralized tracking control of multi-AUV formation

4.2.7 Nonlinear cross-track control of an underactuated AUVs

The stabilization function of the yaw angle is designed to stabilize the cross-track error as the virtual input, resulting in the cascaded subsystems of the cross-track and the yaw tracking as shown in Fig. $30^{[40]}$. For the normal cross-track subsystem when the yaw and yaw rate tracking errors are zeros, the control parameter condition is derived to make the cross-track error and the sway velocity globally asymptotically stable^[96].



Fig. 30 Nonlinear cross-track control of an underactuated AUVs

$$\psi LOS = \arctan \frac{Y_{way(i)} - y(t)}{X_{way(i)} - x(t)}.$$
(15)

Suppose that the current navigation angle of AUV is $\psi(t)$, the position in inertial frame is $(x(t), y(t), \psi_d(i))$, where $\psi_d(i)$ is the angle of the current line of tracking^[96], i.e.,

$$\psi_d^i = \arctan \frac{Y_{way(i)} - Y_{way(i-q)}}{X_{way(i)} - X_{way(i-q)}}.$$
 (16)

The navigation tracking error is given by^[96]

$$\psi(t)_{CTE(i)} = \psi_d(i) - \psi(t). \tag{17}$$

The distance from current AUV to the next way point ${\rm is}^{[96]}$

$$S(t)_{i} = \sqrt{X(t)_{way}^{2}(i) + Y(t)_{way(i)}^{2}}$$
(18)

where $X(t)_{way(i)} = X_{way(i)} - x(t)$ and $Y(t)_{way(i)} = Y_{way(i)} - y(t)^{[96]}$.

Then, the cross tracking error $\varepsilon(t)$ is given by

$$\varepsilon(t) = S(t)_i \sin(d_p(t)) \tag{19}$$

where

$$d_p(t) = \psi_{LOS} - \psi_d(i). \tag{20}$$

4.2.8 Nonlinear formation-keeping and mooring control of multiple AUVs

A nonlinear formation keeping and mooring control of multiple AUVs is proposed in [97]. The AUV formation under consideration is constrained by the desired separations and orientations of follower AUVs with respect to a time-varying leader AUV as shown in Fig. 31^[98]. The process uses as follows.

1) A time-varying, smooth feedback control law for the formation-keeping of multiple non-holonomic AUVs is proposed.

2) A time-varying, smooth feedback control law with asymptotic stability is designed to collaboratively moor the follower AUV to its desired docking position and orientation with respect to the leader by using the integrator back stepping method.

3) The realization problems of physical AUV system and singularity avoidance are investigated for applying the aforementioned control laws to a real formation system of AUVs.



Fig. 31 Schematic of the leader–follower formation of $\rm AUVs^{[97, \, 98]}$

4.2.9 Finite-time consensus algorithms for multiple AUVs

Based on homogeneous control method, finite-time consensus algorithms are proposed for both leaderless and leader-follower multi-AUV formation^[99, 100]. In the leaderfollower case, a distributed finite-time observer is developed for the followers to estimate the leader's velocity. In the cooperative control problems of multi-agent systems, all the agents reach the agreement on a common state by implementing appropriate consensus protocols. Due to the above superiorities and the role of the consensus problem in distributed cooperative control field, several kinds of finitetime consensus algorithms have been developed for both first-order and second-order multi-agent systems recently as shown in Fig. $32^{[101]}$.



Fig. 32 Coordination of AUVs through distributed approach

4.2.10 Range-based formation control

In this formation only ranges are obtained from the leading AUVs without knowledge of the formation path shown in Fig. 33.

4.2.11 Hierarchical control system

The particular AUVs used for the surveillance task are equipped with direction finding sensors and communication

payloads. One particular motivation for the research is accurate cooperative localization of radar systems with a small AUV fleet $^{[102, 103]}$.

In a hierarchical structure, control tasks are classified and distributed non-uniformly among agents. A commonly used hierarchical structure within the formation control literature is the leader-follower structure. In the leaderfollower structure, one "leader" agent is provided with direction and/or path information and is responsible for path tracking, manoeuvring and guiding tasks. The other "follower" agents within the formation measure their distances and/or bearings to a set of "leader" or "follower" agents or both, and are usually required to maintain the shape of the formation via keeping certain fixed distances as shown in Fig. $34^{[50, 104]}$.



Fig. 33 System of three AUVs and their intended triangular formation



Fig. 34 Network topology for a group of agents with a virtual leader

4.2.12 Passivity based approach

In this section, the formation control problem of AUV under the passivity-based group coordination framework is proposed in [105]. A consensus tracking approach is also proposed. The desired formation patterns are obtained when the common reference velocity assigned by a dynamic virtual leader is available to only one AUV or one subset of AUVs^[106]. A comparison study is made on various trajectory tracking based formation control strategies as provided in Table 2.

210

		Table 1	Comparison of	various regulatory formation control sti	trategies	
Formation technique	Control strategy	Control variable	Complexity level	Advantage	Disadvantage	Applications
Gateway coordination	Feed-back linearization	$\operatorname{Yaw}(\psi)$	Simple	Knowledge of each AUV's position makes the formation keeping easier	If any one AUV gets damaged, the whole formation gets affected	It is used for monitoring pipeline surveillance
Neighbour's reference	Decentralized co-operative control technique	Position vector (η)	Simple	Each AUV communicates to its immediate neighbours for coordination	Communication delays between neighbours are inevitable	Coordination of a school of robotic fish
Centralized approach	Centralized cooperative control technique	Position vector (η) velocity vector	Complex	Power efficient as single computer or less number of sensors are used for monitoring	Communication delays are inevitable High computational power and not robust	Naval mining search operations
NN based dynamic surface control	Adaptive control technique	Position vector (η)	Complex	Reduces the equipment cost and the network burden	Affected by environmental disturbances Dependence on a single	Underwater survey Used in arctic region research
Followers within a cluster	Centralized control	Position vector (η)	Simple	AUVs communicate among themselves	controller Communication delays are inevitable	Formation of mobile AUVs
Spatially synchronized parallel formation	Co-operative synchronized control	Velocity vector	Simple	It is a parallel formation Affected by communication delays	Difficult to discover AUV if any one of the vehicles gets damaged	Distance fixing and observation synchronization
Non-hierarchical control	Periodic formation control	position vector (η)	Simple	Distance maintaining and path tracking are uniformly distributed due to agents	Communication delays are inevitable	Underwater survey collision and obstacle avoidance
Stationary surveillance with formation change	Adaptive control	Phase waves phase gradient	Complex	Formation shape may be switched	Communication delays are inevitable	Underwater survey
Cross-track formation	Line of sight guidance	Velocity	Simple	Flexible and easier	Communication delays are inevitable	Collision and obstacle avoidance
Geometric formation	Velocity optimization	Velocity & position	Simple	Very easy Practicable	High computational power	Distance fixing and observation synchronization
Boundary-based geometric formation	Region boundary technique	Position vector (η)	Simple	Effective to carry out individual task.	High computational power	Edge-based segmentation approach is used for pipeline inspection.
Behavioural approach	Behavioural technique	Coordinate (η)	Complex	Guide AUVs in an unknown and uncertain environment	Mathematically more complex Not much practicable	Goal seeking Obstacle avoidance
Genetic algorithm	Path planning technique	Yaw (ψ)	Complex	Overcome disadvantages of memory consumption and Octree approach	High computational power	Practical aquatic survey Studying underwater environment

Table 1 Comparison of various regulatory formation control strategies

Application	Underwater survey	Aquatic research work	Swarm mobilization of AUVs	Naval operations and underwater missions using multiple under actuated AUVs	Underwater tracking activities Underwater mine detection	Arctic missions Deployment of underwater missiles	Used in establishing proper docking position and orientation	Used in first order and second order multi-agent systems (MAS)	Path tracking Maneuvering and guiding tasks	Small AUV fleet formation control Tracking activities
Disadvantage	No feedback from the followers	Lack of robustness	Leader is a single point of failure	Difficult to design a controller such that all the optimized points except for the desired equilibrium one are unstable	Unable or extremely difficult to predict and control the critical points Very sensitive to external disturbance	Difficult to predict and control the critical points	Neighbour to neighbour communication is required	Underwater communication is prior issue which is complex	Very sensitive to external disturbance Basically model based	Very difficult to solve any formation problem by individual AUV in the group
Advantage	Relatively easy to understand and implement	Formation can still be maintained even if the leader is perturbed by some disturbance	Reduces the total huge task to individual small tasking problems	Independent of geometry of formation No chance of fault in existing or switching on formation shapes	Reduces the controller's computational power Independent of any central controller Effective when the communication is limited	Easy to implement this controller Independent of geometry formation	AUV only transmits state information to its neighbour vehicle	Effective when the communication among the AUVs is limited	Communication is usually asynchronous and locally transmitted	Possible to avoid the problem caused by multiple packet transmission
Complexity level	Simple	Simple	Simple	Simple	Simple	Simple	Complex	Simple	Simple	Simple
Control variable	Position vector (η)	Position vector (η)	Yaw (ψ)	Position vector (η)	Position vector (η)	$\operatorname{Yaw}(\psi)$	Yaw (ψ) velocity	Position vector (η) velocity	Position vector (η)	velocity
Control strategy	Decentralized controller	Decentralized controller	Fuzzy Logic based controller	Decentralized back-stepping technique	Expansion and contraction paradigm	Decentralized control technique	Back-stepping control technique	Consensus based technique	Accurate cooperative localization	Consensus tracking approach
Formation technique	Virtual leader follower hierarchy structure	Coordinated control in the presence of communication losses	Fuzzy logic based behaviour	Synchronized path following control	Decentralized overlapping tracking	Nonlinear cross-track control	Nonlinear formation-keeping and mooring control	Finite-time consensus algorithms for AUVs	Range based formation control	Passivity based tracking

Table 2 Comparison of various trajectory tracking based formation control strategies

212

International Journal of Automation and Computing 13(3), June 2016

5 Challenges in formation control

There are several challenges associated with formation control of multiple AUVs. These are classified as wave disturbance, communication constraints, collision and obstacle and may be affecting trajectory tracking, path following, formation shape generation, switching between shapes and path generation for multiple AUVs^[107]. These are briefly explained below.

5.1 Environmental disturbance

For an AUV to operate with a high degree of reliability, disturbances and their effects on the AUV must be modeled such that adequate degree of accuracy must be achieved^[22]. The main sources of the dynamic disturbances encountered by AUVs are wave and current induced disturbances. The design of AUV model in presence of wave disturbance forces in shallow water is necessary to generate a dynamic model representing the wave induced water velocity and acceleration^[107]. Underwater external sea current disturbances are considered to be external factors causing a cross tracking error. To develop such model, the major effects of oceanic currents on dynamical model of AUV in coastal areas come from tidal currents and Stokes's drift effects^[108].

5.2 Communication constraints

A wireless network is necessary for each AUV to keep the information of its neighbor AUVs as well as to maintain global communication^[109]. While moving in formation, the AUVs should communicate with each other through a modulator/demodulator(MODEM) fitted in AUV according to the control and coordination strategies of the $system^{[110]}$. Within communication range, if the two AUVs are locally communicated then perfect communication channel exists directly but the communication among them is possible by observing the states of the AUVs then this type of communication is non line of sight (NLOS) communication^[111]. If an AUV can communicate with any other AUV within communication range, then it is called global communication. Positions of the AUVs within formation are determined according to desired formation shape based on communication topology^[112]. For AUV communication, acoustic signals may be used without using electromagnetic waves. The chances of multipath propagation between the AUVs are due to disturbances of sea layers, small bandwidth, and strong attenuation of signal in underwater medium, and high latency due to the low speed (1500 m/s) of sound in water^[113]. Thus the data transfer rate is very low in under sea area. During motion of a formation group, when information flows from one AUV to another, there may be chance of data packet loss and/or dropout due to attenuation in the environment and scattering of the information wave in the surrounding. So there is a chance of occurrence of delay. Thus delay compensators can be employed while designing formation control strategies. In formation control of AUVs, graph theory is necessary for communication of different AUVs with each other^[114]. Here, the vertices are placed at the position of AUVs and the links are formed by the directed vectors from one AUV to another^[115]. To overcome the limitations of underwater environment and to create the efficient cooperation among the multiple AUVs, a special type of system DELPHIS may be used^[116,117]. The control and communication based on multimode operation may be used to build up a prototype^[118]. To keep the communication among the fleet of cooperative AUVs, an inter-vehicular communication system may be used with environmental study^[119]. Each node uses more communication range as much as possible. Network algorithm based on time-scheduled operation may be developed to locate the AUV in underwater^[120]. The localization of AUVs is possible by measuring the inter-vehicular propagation delay and by exchanging localization map within water. To use efficiently the low data rate in acoustic communication for AUVs, compact control language $(CCL)^{[121]}$, or linear quadratic (LQ) optimal control are employed where the control algorithm based on onboard computing power^[122, 123] may be used. Navigation problems of AUVs can be solved to some extent by using the acoustic transponder navigation systems. It is cost effective with additional advantages of global positioning systems (GPS).

In case of underwater networks, routing design for adhoc routing for wireless radio networks, is still being a challenging issue. Distributed protocols are proposed for both delay-sensitive and delay-insensitive applications and allow nodes to select the next hop with the objective of minimizing the energy consumption while taking into account the specific characteristics of acoustic propagation as well as the application requirements^[115, 119]. A geographical approach is also proposed by Leonard et al., where a theoretical analysis has shown that it is possible to identify an optimal path that the mobile nodes may try to achieve by minimizing the total path energy consumption^[1]. Other approaches include pressure routing, where decisions are based on depth, which can be easily determined locally by means of a various sensors^[109].

5.3 Collision and obstacle

When a group of multiple AUVs move in formation, it is necessary to avoid collision between themselves as well as avoid collision with the solid obstacles intersecting the formation path to be travelled by the group^[124].

Obstacle avoidance is highly essential issue in formation control. The obstacle may be static or dynamic^[125]. For collision avoidance between AUVs, there must exist a repulsive force between them^[126]. Similarly, a repulsive force should be established between AUV and the obstacle such that there is no collision between the static obstacles appearing on the desired path as shown in Fig. 35. AUVs should avoid collision with the obstacle and should keep a $\lim_{t\to\infty} \|\eta - \eta_{obs}\| = d_s \text{ safe distance from the obstacle. Where } \eta_{obs} = [x_{obs}, y_{obs}, \psi_{obs}]^{\mathrm{T}}$ is the position of the obstacle, d_s is the safest distance of the AUV form obstacle. Some of the best methods for avoiding collision between an AUV,



Fig. 35 Line-of-sight and collision avoidance path $components^{[125]}$

Inertial reference

its neighbors and obstacles are discussed below.

5.3.1 Artificial potential function based approach

In this method, a repulsive potential function between AUV and obstacle is developed which is inversely proportional to the norm of the distance between them^[127, 128]. So when the distance between the AUV and the obstacle decreases, the repulsive force increases and vice versa^[129]. To design the obstacle avoidance algorithm, the maneuvering area is divided into three zones: safe zone, avoidance zone and danger zone. The total potential energy is divided into two different parts: One is potential energy between the two AUVs and the other is potential energy between the vehicle and the target.

5.3.2 Soft computing based approach

Mixed integer quadratic programming (MIQP) optimization method may be used to detect and avoid collision between the vehicles^[130, 131]. This paper describes a heuristic search technique carrying out collision avoidance for autonomous underwater vehicles (AUVs). In [132], the search technique adopts fuzzy relational products to conduct pathplanning of intelligent navigation system. For verification, it is compared with A* search method through simulation time, the optimization of path and the amount of memory usage. Bui and Kim^[133] proposes a new heuristic search technique for obstacle avoidance of autonomous underwater vehicles which uses the Bandler and Kohout's-product of a fuzzy relation with a sonar partitioned into seven sections, this method enables AUVs to navigate safely through the obstacle to the goal with the optimal path.

The fuzzy relation between the sonar sections and the properties of a real-time environment is used as a core concept. In [134], a new-style fuzzy inference controller is proposed to address the mentioned problems for AUV path planning and hence a moving obstacle avoidance strategy is developed. In order to get precise fuzzy membership functions, PSO algorithm with strong global optimization ability is employed to tune them, then improve the performance of the proposed fuzzy controller and achieve a reasonable shorter and smoother path to the target^[135].

Fuzzy logic and artificial neural network based controller

is used in [136] to control an autonomous underwater vehicle to avoid obstacles as shown in Fig. 36. The controller can adjust itself to the variations of oceanic environment. In [137, 138], a collision avoidance algorithm is presented based on principles of reinforcement learning and also motion characteristics of an AUV system. Here, a stochastic real value reinforcement learning algorithm for learning functions with continuous outputs is proposed. Here, the obstacle avoidance mission is divided into two parts, i.e., targeting and avoiding behaviour.

5.3.3 Virtual potential based collision avoidance

In [139], a navigation algorithm is presented, which integrates virtual force concept with a potential-field-based method to maneuver AUV in unknown or unstructured environments. The study focused on the free local minimum in potential-field based navigation.



Fig. 36 Fuzzy controller based tracking control to avoid obstacles $^{\left[136\right] }$

5.3.4 A real-time obstacle avoidance using a multibeam forward looking sonar

In [140], a real-time obstacle-avoiding expert AUV system based on a multi-beam forward looking sonar is presented. Expert system is designed based on the information of task execution. The inference engine is designed and implemented according to the images of sonar, which can send algorithm for obstacle avoidance and path replanning^[141-145].

5.3.5 DVZ approach

In [146], a theoretical study of the coordination of the geometrical movements of AUVs formation following a trajectory forming a desired geometrical structure is presented. It is possible to compute the deformation between the real formation of AUVs and a desired formation which helps us to compute the movement that every vehicle should make to obtain zero deformation. An approach namely deformable virtual zones (DVZ) is used in [147] to resolve all the constraints in a homogeneous way and supply a command vector for each vehicle.

5.3.6 Mission planning under dynamic obstacle avoidance

The proposed method is an application of the Markovdecision-process (MDP) based motion planning method for managing both the kinematics and dynamics of an AUV affected by sea flow and obstacles^[148]. The real-time obstacle avoidance needs preplanning when a new obstacle is discovered so that the AUV can find a suitable path around concave obstacles^[149].

5.3.7 Limit cycle process

A method based on limit cycle process may be employed for avoiding obstacles by generating trajectories of the robot manipulators^[150]. Here the shapes of complex obstacles are modeled by unstable limit cycles. The obstacles may be avoided using the same limit cycle method but in a different fashion, as in [151]. Here the trajectories of the obstacles are presented as a set of transitional trajectories which are considered as the solutions of the differential equations presenting a stable limit cycle of elliptical shape. These ellipses encircle the obstacles. When an obstacle is detected on the desired path, the new trajectories of the vehicle are generated satisfying those differential equations to avoid the obstacle. When the obstacle is avoided, the vehicle again returns to the original path. By combining this avoidance strategy of individual obstacle, a group of AUVs in formation can avoid obstacles during travelling along the desired $trajectories^{[152, 153]}$.

Energy problem in the case of most AUVs is a traditional problem. The batteries which provide energy contain silver-zinc composition or lead-acid composition. But now commercial nickel metal hydride (Ni-MH) batteries are available which can provide more energy. Another way of solving this problem is using of solar cells as supplementary energy sources. This can increase the endurance of the energy cells.

6 Formation control stability

Assumption 1. The dynamics equations (5)-(12) of AUV provides the following properties^[1, 154-162].

1) M_{η} is symmetric positive stable for any number of AUVs, $i \in (1, 2, 3, \cdots, N)$

$$\lambda_{\min}(M_i) \|x\|_2^2 \le x^{\mathrm{T}} M_{\eta}(\eta_i) x \le \lambda_{\max}(M_i) \|x\|_2^2, \ \forall x \neq 0.$$
(21)

2) C_{η} is a skew symmetric and which can be termed as^[1]

$$x^{\mathrm{T}}\left(\frac{1}{2}\dot{M}_{\eta}(\eta_{i}) - 2C_{\eta}(\eta_{i},\dot{\eta}_{i})\right)x = 0, \ \forall x \in \mathbf{R}^{3}.$$
 (22)

3) The damping matrix $D_{\eta}(\eta_i, \dot{\eta}_i)$ is positive such that^[1]

$$x^{\mathrm{T}} D_{\eta}(\eta_i, \dot{\eta}_i) x > 0, \ \forall x \neq 0.$$
⁽²³⁾

4) Embedding a load will not only change the mass matrix, but will also induce torques as the gravity and buoyancy centers will not coincide anymore and hence g will change^[1]. But as it is considered for three DOF of motion along (x, y) axis, hence $g(\eta_i) = 0$. 5) Centre of mass and centre of buoyancy coincide with each other and other terms such as roll motions and hydrodynamic terms of higher order are assumed to be negligible^[1].

Formation control is the control of interconnected system of multiple AUVs. For safety, robustness as well as for getting desired performance from an interconnected system, stability analysis of the system is necessary. This is essential to make a system operationally stable. There are three stability notions of formation control. These are string stability, mesh stability and leader-to-formation stability which are interconnected to each other.

Assumption 2. An interconnected nonlinear AUV system is called look-ahead, if the (i, j)-th subsystem is connected only to the subsystems (k, l) such that $k \leq i$ and $l \leq j$. The look-ahead condition may be defined as in [163–172].

Consider a system of the $form^{[173]}$

$$\dot{\eta}_i = f_i \left(\eta_i, \eta_{i-1}, \cdots, \eta_N \right) \tag{24}$$

where $i \in (1, \dots, N)$, $\eta_i \in \mathbf{R}^n$, $f : \mathbf{R}^n \times \dots \times \mathbf{R}^n \to \mathbf{R}^n$ and $f(0, \dots, 0) = 0$.

Assumption 3. Consider the state ξ of leader-follower formation and the formation performance output z. The formation state is obtained from the original state vectors of the AUV and controllers by a coordinate transformation^[173-183].

$$\xi = \chi(t, z) \tag{25}$$

where $\chi \stackrel{\Delta}{=} (\chi_1, \chi_2, \dots, \chi_N)$. The performance output z is a function of the formation state. The closed loop formation dynamics then are given by

$$\dot{\xi} = f(t,\xi,d)$$

$$z = h(t,\xi)$$
(26)

where $f(0, \dots, 0) = 0$ and h(t, 0) = 0 for all t. The following conditions for formation (25) must be satisfied, if

1) the system (25) has well defined solution for all $t \ge 0$, with initial conditions $\xi(t_0)$;

2) all solutions of (24) must satisfy

$$\|z\|_{\infty} \leq \max\left\{\rho\left(\|\xi\left(t_{0}\right)\|\right), \lambda\left(\|d\|_{\infty}\right)\right\}$$

$$\lim_{x \to \infty} \sup\|z\left(t\right)\| \leq \lambda\left(\lim_{x \to \infty} \sup\|d\left(t\right)\|\right)$$
(27)

where $\|z\|_{\infty} \stackrel{\Delta}{=} \sup_{t>0} \|z(t)\|.$

Assumption 4. This type of stability criteria are used to check the stability of the systems operating in platoon structures or hybrid platoon system which obeys the lookahead conditions. This stability method can be applicable to both constrained and unconstrained systems. The stability criteria are applied in an array of linear interconnected AUVs under communication constraint condition in [77, 183–192]. String stability presents the uniform boundedness of the states of all interconnected systems. Mesh stability is used for ensuring stability property by considering attenuation of error of multiple interconnected AUVs. The origin $\eta_i = 0, i \in N$ of (23) is string stable if for a given $\in>0,$ there exists a $\delta>0$ such that the following condition is satisfied $^{[173]}$

$$\left\|\eta_{i}\left(0\right)\right\|_{\infty} < \delta \Rightarrow \sup\left\|\eta_{i}\left(\cdot\right)\right\|_{\infty} < \epsilon .$$

$$(28)$$

The origin $x_i = 0$, $i \in N$, of (23) is exponentially string stable if it satisfies the condition of string stability and $\eta_i(t) \to 0$ asymptotically, $\forall i \in N$. For a step change in \dot{q}_L at any time t = 0, the leader-follower interconnected system will be asymptotically stable if the following condition is satisfied. For every $j = 2, \dots, N_v$, there exists a constant $\alpha_j \in (0, 1)$ so that the following closed-loop position error satisfies^[173]

$$\max_{t \ge 0} |q_{j,e}(t)| \le \alpha_j \max_{t \ge 0} |q_{1,e}(t)|, \forall j = 2, \cdots, N_v.$$
(29)
Leader-follower string stability

 $\max_{t\geq 0} |q_{j,e}(t)| \leq \alpha_j \max_{t\geq 0} \left| q_{(j-1),e}(t) \right|, \, \forall j = 2, \cdots, N_v. (30)$ Predecessor-follower string stability

$$\max_{t > 0} |q_{j,e}(t)| \le \alpha_j \max_{t > 0} |q_{(j-1),e}(t)|, \, \forall j = 2, \cdots, N_v. \, (31)$$

 q_L and \dot{q}_L are the geometric position and speed of AUVs. For Mesh stability, the dynamical system (23) is globally exponentially mesh stable if the following conditions are satisfied, given $\in > 0$, there exists a $\delta > 0$ such that^[173]

$$\left\| x\left(0\right) \right\|_{\infty} < \delta \Rightarrow \left\| x\left(t\right) \right\|_{\infty} < \in \tag{32}$$

$$x \to 0$$
, exponentially $\forall x \in \mathbf{R}^n$ (33)

$$\|x_i(t)\|_{\infty} \le \|x_i(t)\|_{\infty}^{i-1}, \forall i \in \{2, \cdots, N\}.$$
 (34)

7 Applications of formation control of AUVs

Most of the applications of AUV require that it should follow a desired path or surveillance of a desired region. AUVs are now being used for tasks with roles and missions based on navigation system, guidance system and control structure as shown in Fig. 37.



Fig. 37 Applications of multiple AUVs in formation^[32, 33]

7.1 Guidance system

The guidance system helps to determine the path generation from AUVs with the help of current position to the desired position. The feasible path will differ for different AUVs depending on fully-actuated or underactuated system. It is necessary for the AUV to follow the desired path successfully considering the obstacles across the paths^[34, 193–202].

7.2 Control structure of AUV system

Control structure determines the required control forces necessary for steering the AUV along the desired path^[203-205]. Control structure based on requirement is divided into trajectory tracking, path following and way point tracking^[206-208]. While developing a control law, it is necessary to check the generated control forces which should reside within desired limit^[209-211]. The motion control strategies to accomplish the mission of AUVs are classified as trajectory tracking, path following and way point tracking^[35, 212-215].

AUVs are employed in missions such as oceanographic observations, bathymetric surveys, ocean floor analysis, military applications, recovery of lost objects, etc^[36]. The applications of underwater vehicles have shown a dramatic increase in recent years, such as mines clearing operation, feature tracking, cable or pipeline tracking and deep ocean exploration. The mission areas include commercial applications by surveying of the sea floor for oil and gas industry, mine countermeasures, monitoring and safeguarding protected areas and oceanic research. AUVs are also employed for military purpose such as anti-submarine warfare, to aid in the detection of manned submarines^[216].

8 Conclusions

For decades, AUVs have been widely used for many tasks. The ruthless and unstructured nature of the underwater environment causes significant challenges for underwater autonomous systems. This paper presents a comprehensive review on the current control issues on a group of AUVs. For decades, formation control has become an active research topic and has broad applications in Robotics. The formation control algorithms are subdivided based on the technical approach, controllers used, level of coordination and communication constraints. The paper has also highlighted some areas for future work in the field. Recent advances in stability analysis have been developed for rapid improvement of formation control. In addition, some basic challenges and applications have been presented. The paper also informs briefly a new consideration on formation control stability which has a promising future research direction.

References

 T. I. Fossen. Guidance and Control of Ocean Vehicles, New York, USA: Wiley, 1994.

- [2] J. Yuh. Design and control of autonomous underwater robots: A survey. Autonomous Robots, vol. 8, no. 1, pp. 7– 24, 2000.
- [3] H. Kondo, K. Nakane, E. Shimizu, J. K. Choi, K. Nagahashi, M. Matsushima, Y. Nishida, T. Arimoto, Y. Miyamoto, K. Amakasu, M. Endo, R. Matsui. Design and concept of a biointeractive autonomous underwater vehicle "BA-1". In *Proceedings of IEEE OCEANS*, IEEE, Sydney, Australia, pp. 24–27, 2011.
- [4] G. N. Roberts, R. Sutton. Advances in Unmanned Marine Vehicles, London, UK: The Institution of Engineering and Technology, 2006.
- [5] D. Horner, O. Yakimenko. Recent Developments for an Obstacle Avoidance System for a Small AUV, Monterey, USA: Naval Postgraduate School, 2007.
- [6] J. W. Nicholson, A. J. Healey. The present state of autonomous underwater vehicle (AUV) applications and technologies. *Marine Technology Society Journal*, vol. 42, no. 1 pp. 44–51, 2008.
- [7] D. J. Stilwell, B. E. Bishop. Platoons of underwater vehicles. *IEEE Control Systems Magazine*, vol. 20, no. 6, pp. 45– 52, 2000.
- [8] Y. T. Wang, W. S. Yan. Path parameters consensus based formation control of multiple Autonomous Underwater Vehicles in the presence of ocean currents. In Proceedings of the 17th International Conference on Methods and Models in Automation and Robotics, IEEE, Miedzyzdrojie, Poland, pp. 427–432, 2012.
- [9] Z. H. Peng, D. Wang, Z. Y Chen, X. J. Hu, W. Y. Lan. Adaptive dynamic surface control for formations of autonomous surface vehicles with uncertain dynamics. *IEEE Transactions on Control Systems Technology*, vol. 21, no. 2, pp. 513–520, 2013.
- [10] H. B. Duan, Q. N. Luo, Y. H. Shi, G. J. Ma. Hybrid particle swarm optimization and genetic algorithm for multi-UAV formation reconfiguration. *IEEE Computational Intelligence Magazine*, vol. 8, no. 3, pp. 16–27, 2013.
- [11] X. H. Wang, V. Yadav, S. N. Balakrishnan. Cooperative UAV formation flying with obstacle/collision avoidance. *IEEE Transactions on Control System Technology*, vol. 15, no. 4, pp. 672–679, 2007.
- [12] J. M. Soares, A. P. Aguiar, A. M. Pascoal, A. Martinoli. Joint ASV/AUV range-based formation control: Theory and experimental results. In *Proceedings of IEEE International Conference on Robotics and Automation*, IEEE, Karlsruhe, Germany, pp. 5579–5585, 2013.
- [13] B. K. Sahu, B. Subudhi. Adaptive tracking control of an autonomous underwater vehicle. *International Journal of Automation and Computing*, vol. 11, no. 3, pp. 299–307, 2014.
- [14] Y. Q. Chen, Z. M. Wang. Formation control: A review and a new consideration. In Proceedings of IEEE/RSJ International Conference on Intelligent Robots and Systems, IEEE, Edmonton, Canada, pp. 3181–3186, 2005.

- [15] L. Paull, S. Saeedi, M. Seto, H. Li. AUV navigation and localization: A review. *IEEE Journal of Oceanic Engineering*, vol. 39, no. 1, pp. 131–149, 2014.
- [16] C. J. Cannell, D. J. Stilwell. A comparison of two approaches for adaptive sampling of environmental processes using autonomous underwater vehicles. In *Proceedings of IEEE OCEANS*, IEEE, Washington, USA, vol. 2, pp. 1514– 1521, 2005.
- [17] J. Almeida, C. Silvestre, A. M. Pascoal. Cooperative control of multiple surface vessels with discrete-time periodic communications. *International Journal of Robust and Nonlinear Control*, vol. 22, no. 4, pp. 398–419, 2012.
- [18] W. Ren. Consensus strategies for cooperative control of vehicle formations. *IET Control Theory & Applications*, vol. 1, no. 2, pp. 505–512, 2007.
- [19] H. Z. Yang, F. M. Zhang. Robust control of formation dynamics for autonomous underwater vehicles in horizontal plane. Journal of Dynamic Systems, Measurement, and Control, vol. 134, no. 3, Article number 031009, 2012.
- [20] B. D. O. Anderson, C. B. Yu, S. Dasgupta, A. S. Morse. Control of a three-coleader formation in the plane. Systems and Control Letters, vol. 56, no. 9–10, pp. 573–578, 2007.
- [21] C. J. F. Silvestre. Multi-Objective Optimization Theory with Applications to the Integrated Design of Controllers/Plants for Autonomous Vehicles, Ph. D. dissertation, Technical Institute of Lisbon, Portugal, 2000.
- [22] D. Kruger, R. Stolkin, A. Blum, J. Briganti. Optimal AUV path planning for extended missions in complex, fastflowing estuarine environments. In *Proceedings of IEEE International Conference on Robotics and Automation*, IEEE, Roma, Italy, pp. 4265–4270, 2007.
- [23] R. Kumar, J. A. Stover. A behaviour-based intelligent control architecture with application to coordination of multiple underwater vehicles. *IEEE Transactions on Systems*, *Man, and Cybernetics*, vol. 30, no. 6, pp. 767–784, 2000.
- [24] J. R. T. Lawton, R. W. Beard, B. J. Young. A decentralized approach to formation maneuvers. *IEEE Transactions* on *Robotics and Automation*, vol. 19, no. 6, pp. 933–941, 2003.
- [25] M. J. Mataric. Behaviour-based control: Examples from navigation, learning, and group behaviour. Journal of Experimental and Theoretical Artificial Intelligence, vol.9, no. 2–3, pp. 323–336, 1997.
- [26] W. L. Huang, H. J. Fang, L. Liu. Obstacle avoiding policy of multi-AUV formation based on virtual AUV. In Proceedings of IEEE International Conference on Fuzzy Systems and Knowledge Discovery, IEEE, Washington, USA, vol. 4, pp. 131–135, 2009.
- [27] X. Kang, H. Xu, X. Feng. Fuzzy logic based behavior fusion for multi-AUV formation keeping in uncertain ocean environment. In *Proceedings of IEEE Biloxi-Marine Technology* for Our Future: Global and Local Challenges, IEEE, Biloxi, USA, pp. 1-7, 2009.

- [28] K. D. Do. Formation tracking control of unicycle-type mobile robots with limited sensing ranges. *IEEE Transactions* on *Control Systems Technology*, vol. 16, no. 3, pp. 527–538, 2008.
- [29] D. Sun, C. Wang, W. Shang, G. Feng. A synchronization approach to trajectory tracking of multiple mobile robots while maintaining time-varying formations. *IEEE Transactions on Robotics*, vol. 25, no. 5, pp. 1074–1086, 2009.
- [30] J. Ghommam, F. Mnif. Coordinated path-following control for a group of under actuated surface vessels. *IEEE Transactions on Industrial Electronics*, vol. 56, no. 10, pp. 3951– 3963, 2009.
- [31] K. Teo, K. W. Ong, H. C. Lai. Obstacle detection, avoidance and anti collision for MEREDITH AUV. In Proceedings of IEEE Biloxi-Marine Technology for Our Future: Global and Local Challenges, IEEE, Biloxi, USA, pp. 1–10, 2009.
- [32] [Online], Available: http://oceanexplorer.noaa.gov / explorations / 08auvfest / background / auvs / auvs.html.
- [33] J. G. Bellingham, C. A. Goudey, T. R. Consi, J. W. Bales, D. K. Atwood, J. J. Leonard, C. Chryssostomidis. A second generation survey AUV. In Proceedings of IEEE Symposium on Autonomous Underwater Vehicle Technology, IEEE, Cambridge, USA, pp. 148–155, 1994.
- [34] L. R. Fodrea. Obstacle Avoidance Control for the RE-MUS Autonomous Underwater Vehicle, Master dissertation, Naval Postgraduate School, Monterey, USA, 2002.
- [35] C. D. Chuhran. Obstacle Avoidance Control in the Vertical Plane for the REMUS Autonomous Underwater Vehicle, Ph. D. dissertation, Naval Postgraduate School, Monterey, USA, 2003.
- [36] J. C. Hyland, F. J. Taylor. Mine avoidance techniques for underwater vehicles. *IEEE Journal of Oceanic Engineering*, vol. 18, no. 3, pp. 340–350, 1993.
- [37] C. W. Chen, J. S. Kouh, J. F. Tsai. Modeling and simulation of an AUV simulator with guidance system. *IEEE Journal of Oceanic Engineering*, vol. 38, no. 2, pp. 211–225, 2013.
- [38] S. Wadoo, P. Kachroo. Autonomous Underwater Vehicles, London, UK: CRC Press, 2011.
- [39] L. Paull, S. Saeedi, M. Seto, H. Li. AUV navigation and localization: A review. *IEEE Journal of Oceanic Engineering*, vol. 39, no. 1, pp. 131–149, 2014.
- [40] Y. T. Wang, W. S. Yan, W. Yan. A leader-follower formation control strategy for AUVs based on line-of-sight guidance. In Proceedings of IEEE International Conference on Mechatronics and Automation, IEEE, Changchun, China, pp. 4863–4867, 2009.
- [41] W. Dong, J. A. Farrell. Formation control of multiple underactuated surface vessels. *IET Control Theory & Applications*, vol. 2, no. 12, pp. 1077–1085, 2008.

- International Journal of Automation and Computing 13(3), June 2016
 - [42] N. Burlutskiy, Y. Touahmi, B. H. Lee. Power efficient formation configuration for centralized leader-follower AUVs control. *Journal of Marine Science and Technology*, vol. 17, no. 3, pp. 315–329, 2012.
 - [43] R. Amin, A. A. Khayyat, K. G. Osgouie. Neural networks control of autonomous underwater vehicle. In Proceedings of the 2nd International Conference on Mechanical and Electronics Engineering, IEEE, Kyoto, Japan, vol. 2, pp. V2-117–V2-121, 2010.
 - [44] C. A. Kitts, I. Mas. Cluster space specification and control of mobile multi-robot systems. *IEEE/ASME Transactions* on Mechatronics, vol. 14, no. 2, pp. 207–218, 2009.
 - [45] M. N. Soorki, H. A. Talebi, S. K. Y. Nikravesh. A robust dynamic leader-follower formation control with active obstacle avoidance. In Proceedings of IEEE International Conference on System, Man, and Cybernetics, IEEE, Anchorage, USA, pp. 1932–1937, 2011.
 - [46] R. Ghabcheloo, A. P. Aguiar, A. Pascoal, C. Silvestre, I. Kaminer, J. Hespanha. Coordinated path-following control of multiple underactuated autonomous vehicles in the presence of communication failures. In *Proceedings of the* 45th IEEE Conference on Decision and Control, IEEE, San Diego, USA, pp. 4345–4350, 2006.
 - [47] E. Borhaug, K. Y. Pettersen. Formation control of 6-DOF Euler-Lagrange systems with restricted inter-vehicle communication. In Proceedings of the 45th IEEE Conference on Decision and Control, IEEE, San Diego, USA, pp. 5718– 5723, 2006.
 - [48] D. van der Walle, B. Fidan, A. Sutton, C. B. Yu, B. D. O. Anderson. Non-hierarchical UAV formation control for surveillance tasks. In *Proceedings of American Control Conference*, IEEE, Seattle, USA, pp. 777–782, 2008.
 - [49] J. A. Marshall, D. Tsai. Periodic formations of multivehicle systems. *IET Control Theory & Applications*, vol. 5, no. 2, pp. 389–396, 2011.
 - [50] L. Brinon-Arranz, A. Seuret, C. Canudas-de-Wit. Collaborative estimation of gradient direction by a formation of AUVs under communication constraints. In Proceedings of the 50th IEEE Conference on Decision and Control and European Control Conference, IEEE, Orlando, USA, pp. 5583– 5588, 2011.
 - [51] L. Brinon-Arranz, A. Seuret, C. C. De Wit. Translation control of a fleet circular formation of AUVs under finite communication range. In Proceedings of IEEE Decision and Control and the 28th Chinese Control Conference, IEEE, China, pp. 8345–8350, 2009.
 - [52] T. Keviczky, F. Borrelli, K. Fregene, D. Godbole, G. J. Balas. Decentralized receding horizon control and coordination of AUV formations. *IEEE Transactions on Control Systems Technology*, vol. 16, no. 1, pp. 19–33, 2008.
 - [53] S. Liu, D. Sun, C. Zhu. Coordinated motion planning for multiple mobile robots along designed paths with formation requirement. *IEEE/ASME Transactions on Mechatronics*, vol. 16, no. 6, pp. 1021–1031, 2011.

218

D Springer

- [54] J. Wang, X. B. Wu, Z. L. Xu. Decentralized formation control and obstacles avoidance based on potential field method. In *Proceedings of International Conference on Machine Learning and Cybernetics*, IEEE, Dalian, China, pp. 803–808, 2006.
- [55] M. Defoort, T. Floquet, A. Kokosy, W. Perruquetti. Slidingmode formation control for cooperative autonomous mobile robots. *IEEE Transactions on Industrial Electronics*, vol. 55, no. 11, pp. 3944–3953, 2008.
- [56] Y. S. Jung, K. W. Lee, S. Y. Lee, M. H. Choi, B. H. Lee. An efficient underwater coverage method for multi-AUV with sea current disturbances. *International Journal of Control*, *Automation, and Systems*, vol. 7, no. 4, pp. 615–629, 2009.
- [57] X. P. Wu, Z. P. Feng, J. M. Zhu, R. Allen. Line of sight guidance with intelligent obstacle avoidance for autonomous underwater vehicles. In *Proceedings of IEEE OCEANS*, IEEE, Boston, USA, pp. 1–6, 2006.
- [58] R. X. Cui, D. M. Xu, W. S. Yan. Formation control of autonomous underwater vehicles under fixed topology. In Proceedings of IEEE International Conference on Control and Automation, IEEE, Guangzhou, China, pp. 2913–2918, 2007.
- [59] F. M. Zhang. Geometric cooperative control of particle formations. *IEEE Transactions on Automatic Control*, vol. 55, no. 3, pp. 800–804, 2010.
- [60] J. Almeida, C. Silvestre, A. Pascoal. Cooperative control of multiple surface vessels in the presence of ocean currents and parametric model uncertainty. *International Journal* of Robust and Nonlinear Control, vol. 20, no. 14, pp. 1549– 1565, 2010.
- [61] F. Dorfler, B. Francis. Geometric analysis of the formation problem for autonomous robots. *IEEE Transactions on Au*tomatic Control, vol. 55, no. 10, pp. 2379–2384, 2010.
- [62] H. Z. Yang, F. M. Zhang. Robust control of horizontal formation dynamics for autonomous underwater vehicles. In Proceedings of IEEE International Conference on Robotics and Automation, IEEE, Shanghai, China, pp. 3364–3369, 2011.
- [63] Z. H. Ismail, M. W. Dunnigan. A region boundary-based geometric formation control scheme for multiple Autonomous Underwater Vehicles. In Proceedings of IEEE International Conference on Electrical, Control and Computer Engineering, IEEE, Pahang, Malaysia, pp. 491–496, 2011.
- [64] Z. H Ismail, N. Sarman, M. W. Dunnigan. Dynamic region boundary-based control scheme for Multiple Autonomous Underwater Vehicles. In *Proceedings of IEEE OCEANS*, IEEE, Yeosu, South Korea, pp. 1–6, 2012.
- [65] H. Z. Yang, C. F. Wang, F. M. Zhang. Robust geometric formation control of multiple autonomous underwater vehicles with time delays. In *Proceedings of American Control Conference*, IEEE, Washington, USA, pp. 1380–1385, 2013.
- [66] H. Z. Yang, F. M. Zhang. Geometric formation control for autonomous underwater vehicles. In Proceedings of IEEE International Conference on Robotics and Automation, IEEE, Anchorage, USA, pp. 4288–4293, 2010.

- [67] L. J. Zhang, X. Qi. Muti-AUV's formation coordinated control in the presence of communication losses. In *Proceedings of the 32nd Chinese Control Conference*, IEEE, Xi'an, China, pp. 3089–3094, 2013.
- [68] X. B. Xiang, C. Liu, L. Lapierre, B. Jouvencel. Synchronized path following control of multiple homogenous underactuated AUVs. *Journal of Systems Science and Complexity*, vol. 25, no. 1, pp. 71–89, 2012.
- [69] X. B. Xiang, C. Liu, B. Jouvencel. Synchronized path following control for multiple underactuated AUVs. In Proceedings of the 29th Chinese Control Conference, IEEE, Beijing, China, pp. 3785–3790, 2010.
- [70] W. S. Yan, R. X. Cui, D. M. Xu. Formation control of underactuated autonomous underwater vehicles in horizontal plane. In Proceedings of IEEE International Conference on Automation and Logistics, IEEE, Qingdao, China, pp. 822– 827, 2008.
- [71] Y. Lan, G. F. Yan, Z. Y. Lin. Synthesis of distributed control of coordinated path following based on hybrid approach. *IEEE Transactions on Automatic Control*, vol. 56, no. 5, pp. 1170–1175, 2011.
- [72] H. Z. Yang, C. F. Wang, F. M. Zhang. A decoupled controller design approach for formation control of autonomous underwater vehicles with time delays. *IET Control Theory* & Applications, vol. 7, no. 15, pp. 1950–1958, 2013.
- [73] X. B. Xiang, B. Jouvencel, O. Parodi. Coordinated formation control of multiple autonomous underwater vehicles for pipeline inspection. *International Journal of Advanced Robotic Systems*, vol. 7, no. 1, pp. 75–84, 2010.
- [74] D. P. Jiang, Y. J. Pang, Z. B. Qin. Coordinated control of multiple autonomous underwater vehicle system. In Proceedings of the 8th World Congress on Intelligent Control and Automation, IEEE, Jinan, China, pp. 4901–4906, 2010.
- [75] J. Evans, Y. Petillot, P. Redmond, M. Wilson, D. Lane. AUTOTRACKER: AUV embedded control architecture for autonomous pipeline and cable tracking. In *Proceedings of IEEE OCEANS*, IEEE, San Diego, USA, vol. 5, pp. 2651– 2658, 2003.
- [76] A. Caiti, A. Munafo. Cooperative distributed behaviours of an AUV network for asset protection with communication constraints. In *Proceedings of IEEE OCEANS*, IEEE, Santander, Spain, pp. 1–6, 2011.
- [77] Z. L. Hu, C. Ma, L. X. Zhang, A. Halme, T. Hayat, B. Ahmad. Formation control of impulsive networked autonomous underwater vehicles under fixed and switching topologies. *Neurocomputing*, vol. 147, pp. 291–298, 2015.
- [78] D. Q. Zhu, H. Huang, S. X. Yang. Dynamic task assignment and path planning of multi-AUV system based on an improved self-organizing map and velocity synthesis method in three-dimensional underwater workspace. *IEEE Transactions on Cybernetics*, vol. 43, no. 2, pp. 504–514, 2013.
- [79] D. R. Yoerger, M. Jakuba, A. M. Bradley, B. Bingham. Techniques for deep sea near bottom survey using an autonomous underwater vehicle. *The International Journal of Robotics Research*, vol. 26, no. 1, pp. 41–54, 2007.

- [80] L. Sorbi, G. P. De Capua, J. Fontaine, L. Toni. A behaviourbased mission planner for cooperative autonomous underwater vehicles. *Marine Technology Society Journal*, vol. 46, no. 2, pp. 32–44, 2012.
- [81] Q. R. Zhang. A hierarchical global path planning approach for AUV based on genetic algorithm. In Proceedings of IEEE International Conference on Mechatronics and Automation, IEEE, Luoyang, China, pp. 1745–1750, 2006.
- [82] Y. Kitamura, T. Tanaka, F. Kishino, M. Yachida. 3-D path planning in a dynamic environment using an octree and an artificial potential field. In *Proceedings of IEEE/RSJ International Conference on Intelligent Robots and Systems*, IEEE, Pittsburgh, USA, vol. 2, pp. 474–481, 1995.
- [83] H. G. Tanner, G. J. Pappas, V. Kumar. Leader-toformation stability. *IEEE Transactions on Robotics and Au*tomation, vol. 20, no. 3, pp. 443–455, 2004.
- [84] N. E. Leonard, E. Fiorelli. Virtual leaders, artificial potentials and coordinated control of groups. In *Proceedings of* the 40th IEEE Conference on Decision and Control, IEEE, Orlando, USA, vol. 3, pp. 2968–2973, 2001.
- [85] Z. H. Zhou, J. Yuan, W. X. Zhang, J. P. Zhao. Virtualleader-follower structure and finite-time controller based cooperative control of multiple autonomous underwater vehicles. In Proceedings of the 24th IEEE Chinese Control and Decision Conference, IEEE, Taiyuan, China, pp. 3670–3675, 2012.
- [86] X. Kang, H. Xu, X. Feng. Fuzzy logic based behaviour fusion for multi-AUV formation keeping in uncertain ocean environment. In Proceedings of MTS/IEEE Biloxi-Marine Technology for Our Future: Global and Local Challenges, IEEE, Biloxi, USA, pp. 1–7, 2009.
- [87] W. Zhang, J. H. Hu. Optimal multi-agent coordination under tree formation constraints. *IEEE Transactions on Au*tomatic Control, vol. 53, no. 3, pp. 692–705, 2008.
- [88] S. Fan, Z. Feng, L. Lian. Collision free formation control for multiple autonomous underwater vehicles. In *Proceed*ings of *IEEE OCEANS*, IEEE, Sydney, Australia, pp. 1–4, 2010.
- [89] L. Lapierre, B. Jouvencel. Robust nonlinear path-following control of an AUV. *IEEE Journal of Oceanic Engineering*, vol. 33, no. 2, pp. 89–102, 2008.
- [90] P. Batista, C. Silvestre, P. Oliveira. A sensor-based controller for homing of underactuated AUVs. *IEEE Transactions on Robotics*, vol. 25, no. 3, pp. 701–716, 2009.
- [91] J. E. Refsnes, A. J. Sorensen, K. Y. Pettersen. Modelbased output feedback control of slender-body underactuated AUVs: Theory and experiments. *IEEE Transactions* on Control Systems Technology, vol. 16, no. 5, pp. 930–946, 2008.
- [92] M. F. Fallon, J. Folkesson, H. McClelland, J. J. Leonard. Relocating underwater features autonomously using sonarbased SLAM. *IEEE Journal of Oceanic Engineering*, vol. 38, no. 3, pp. 500–513, 2013.

[93] L. Paull, M. Seto, J. J. Leonard. Decentralized cooperative trajectory estimation for autonomous underwater vehicles. In Proceedings of IEEE/RSJ International Conference

on Intelligent Robots and Systems, IEEE, Chicago, USA,

- [94] M. Barisic, Z. Vukic, N. Miskovic. Decentralized control functions in trajectory guidance of a non-holonomic AUV. In Proceedings of the 18th Mediterranean Conference on Control & Automation, IEEE, Marrakech, Morocco, pp. 1194–1199, 2010.
- [95] N. V. Nagul. A proportional distribution of AUVs under decentralized control. In Proceedings of the 36th International Convention on Information & Communication Technology Electronics & Microelectronics, IEEE, Opatija, Croatia, pp. 1043–1048, 2013.
- [96] E. Borhaug, A. Pavlov, K. Y. Pettersen. Cross-track formation control of underactuated autonomous underwater vehicles. Lecture Notes in Group Coordination and Cooperative Control, Berlin Heidelberg, Germany: Springer, vol. 336, pp. 35–54, 2006.
- [97] E. F. Yang, D. B. Gu. Nonlinear formation-keeping and mooring control of multiple autonomous underwater vehicles. *IEEE/ASME Transactions on Mechatronics*, vol. 12, no. 2, pp. 164–178, 2007.
- [98] E. Borhaug, A. Pavlov, K. Y. Pettersen. Straight line path following for formations of underactuated underwater vehicles. In Proceedings of the 46th IEEE Conference on Decision and Control, IEEE, New Orleans, USA, pp. 2905–2912, 2007.
- [99] A. Caiti, K. Grythe, J. M. Hovem, S. M. Jesus, A. Lie, A. Munafo, T. A. Reinen, A. Silva, F. Zabel. Linking acoustic communications and network performance: Integration and experimentation of an underwater acoustic network. *IEEE Journal of Oceanic Engineering*, vol. 38, no. 4, pp. 758–771, 2013.
- [100] S. H. Li, X. Y. Wang. Finite-time consensus algorithms for multiple AUVs. In Proceedings of the 32nd IEEE Chinese Control Conference, IEEE, Xi'an, China, pp. 5747–5752, 2013.
- [101] S. H. Li, X. Y. Wang. Finite-time consensus and collision avoidance control algorithms for multiple AUVs. Automatica, vol. 49, no. 11, pp. 3359–3367, 2013.
- [102] I. A. F. Ihle, R. Skjetne, T. I. Fossen. Nonlinear Formation control of marine craft with experimental results. In Proceedings of the 43rd IEEE Conference on Decision and Control, IEEE, Nassau, Bahamas, vol. 1, pp. 680–685, 2004.
- [103] S. Y. Wang, C. W. Zheng. A hierarchical evolutionary trajectory planner for spacecraft formation reconfiguration. *IEEE Transactions on Aerospace and Electronic Systems*, vol. 48, no. 1, pp. 279–289, 2012.
- [104] I. F. Ihle, M. Arcak, T. I. Fossen. Passivity-based designs for synchronized path-following. *Automatica*, vol. 43, no. 9, pp. 1508–1518, 2007.
- [105] Y. Wang, W. Yan, J. Li. Passivity-based formation control of autonomous underwater vehicles. *IET Control Theory & Applications*, vol. 6, no. 4, pp. 518–525, 2012.

International Journal of Automation and Computing 13(3), June 2016

pp. 184–191, 2014.

D Springer

- [106] Y. S. Jung, K. W. Lee, S. Y. Lee, M. H. Choi, B. H. Lee. An efficient underwater coverage method for multi-AUV with sea current disturbances. *International Journal of Control*, *Automation, and Systems*, vol. 7, no. 4, pp. 615–629, 2009.
- [107] A. P. Aguiar, A. M. Pascoal. Dynamic positioning and waypoint tracking of under actuated AUVs in the presence of ocean currents. *International Journal of Control*, vol. 80, no. 7, pp. 1092–1108, 2007.
- [108] A. P. Aguiar, A. M. Pascoal. Dynamic positioning and way-point tracking of underactuated AUVs in the presence of ocean currents. In *Proceedings of the 41st IEEE Conference on Decision and Control*, IEEE, Las Vegas, USA, vol. 2, pp. 2105–2110, 2002.
- [109] M. Chitre, S. Shahabudeen, M. Stojanovic. Underwater acoustic communications and networking: Recent advances and future challenges. *Marine Technology Society Journal*, vol. 42, no. 1, pp. 103–116, 2008.
- [110] B. Benson, Y. Li, R. Kastner, B. Faunce, K. Domond, D. Kimball, C. Schurgers. Design of a low-cost underwater acoustic modem. *IEEE Embedded Systems Letters*, vol. 2, no. 3, pp. 58–61, 2010.
- [111] W. S. Ross, R. S. Kennedy. An Investigation of Atmospheric Optically Scattered Non-line-of-sight Communication Links, USA: Research Laboratory of Electronics Cambridge Massachusetts, 1980.
- [112] J. Heidemann, U. Mitra, J. Preisig, M. Stojanovic, M. Zorzi, L. Cimini. Guest editorial-underwater wireless communication networks. *IEEE Journal on Selected Areas in Communications*, vol. 26, no. 9, pp. 1617–1619, 2008.
- [113] I. S. Kulkarni, D. Pompili. Task allocation for networked autonomous underwater vehicles in critical missions. *IEEE Journal on Selected Areas in Communications*, vol. 28, no. 5, pp. 716–727, 2010.
- [114] X. Z. Cheng, H. N. Shu, Q. L. Liang, D. H. C. Du. Silent positioning in underwater acoustic sensor networks. *IEEE Transactions on Vehicular Technology*, vol. 57, no. 3, pp. 1756–1766, 2008.
- [115] I. F. Akyildiz, D. Pompili, T. Melodia. Underwater acoustic sensor networks: Research challenges. Ad Hoc Networks, vol. 3, no. 3, pp. 257–279, 2005.
- [116] M. Stojanovic, L. Freitag, J. Leonard, P. Newman. A network protocol for multiple AUV localization. In *Proceedings* of *IEEE OCEANS*, IEEE, Biloxi, USA, vol. 1, pp. 604–611, 2002.
- [117] L. Brignone, J. Alves, J. Opderbecke. GREX sea trials: First experiences in multiple underwater vehicle coordination based on acoustic communication. In *Proceedings of IEEE OCEANS*, IEEE, Bremen, Germany, pp. 1–6, 2009.
- [118] T. Shimura, H. Ochi, Y. Watanabe. First experiment result of time-reversal communication in deep ocean. Japanese Journal of Applied Physics, vol. 46, no. 7B, pp. 4956, 2007.
- [119] I. F. Akyildiz, D. Pompili, T. Melodia. Challenges for efficient communication in underwater acoustic sensor networks. ACM Sigbed Review, vol. 1, no. 2, pp. 3–8, 2004.

- [120] J. J. Leonard, A. A. Bennett, C. M. Smith, H. J. S. Feder. Autonomous underwater vehicle navigation. In Proceedings of IEEE ICRA Workshop on Navigation of Outdoor Autonomous Vehicles, Leuven, Belgium, 1998.
- [121] P. L. Kempker, A. C. Ran, J. H. van Schuppen. A formation flying algorithm for autonomous underwater vehicles. In Proceedings of the 50th IEEE Conference on Decision and Control and European Control Conference, IEEE, Orlando, USA, pp. 1293–1298, 2011.
- [122] G. F. Trecate, A. Buffa, M. Gati. Analysis of coordination in multi-agent systems through partial difference equations. *IEEE Transactions on Automatic Control*, vol. 51, no. 6, pp. 1058–1063, 2006.
- [123] E. R. B. Marques, J. Pinto, S. Kragelund, P. S. Dias, L. Madureira, A. Sousa, M. Correia, H. Ferreira, R. Goncalves, R. Martins, D. P. Horner, A. J. Healey, G. M. Goncalves, J. B. Sousa. AUV control and communication using underwater acoustic networks. In *Proceedings of IEEE OCEANS Europe*, IEEE, Aberdeen, UK, pp. 1–6, 2007.
- [124] P. Ogren, N. E. Leonard. Obstacle avoidance in formation. In Proceedings of IEEE International Conference on Robotics and Automation, IEEE, Taipei, China, vol. 2, pp. 2492–2497, 2003.
- [125] M. Soulignac. Feasible and optimal path planning in strong current fields. *IEEE Transactions on Robotics*, vol. 27, no. 1, pp. 89–98, 2011.
- [126] J. Guo, S. W. Cheng, T. C. Liu. AUV obstacle avoidance and navigation using image sequences of a sector scanning sonar. In *Proceedings of IEEE International Symposium on* Underwater Technology, IEEE, Tokyo, Japan, pp. 223–227, 1998.
- [127] Q. Q. Jia, G. Li. Formation control and obstacle avoidance algorithm of multiple autonomous underwater vehicles (AUVs) based on potential function and behaviour rules. In Proceedings of IEEE International Conference on Automation and Logistics, IEEE, Jinan, China, pp. 569–573, 2007.
- [128] H. G. Tanner, G. J. Pappas, V. Kumar. Leader-toformation stability. *IEEE Transactions on Robotics and Au*tomation, vol. 20, no. 3, pp. 443–455, 2004.
- [129] Z. Zeng, K. Sammut, A, Lammas, F. P. He, Y. H. Tang. Efficient path re-planning for AUVs operating in spatiotemporal currents. *Journal of Intelligent Robot Systems*, vol. 79, no. 1, pp. 135–153, 2015.
- [130] T. Kopfstedt, M. Mukai, M. Fujita, O. Sawodny. Formation control for mobile robots in partially known environments using mixed integer programming and fuzzy systems. In Proceedings of International Joint Conference SICE-ICASE, IEEE, Busan, Korea, pp. 1832–1837, 2006.
- [131] P. Rui. Multi-UAV formation maneuvering control based on Q-Learning fuzzy controller. In Proceedings of the 2nd International Conference on Advanced Computer Control, IEEE, Shenyang, China, vol. 4, pp. 252–257, 2010.
- [132] Y. Lee, Y. Kim, L. J. Kohout. An intelligent collision avoidance system for AUVs using fuzzy relational products. *Information Sciences*, vol. 158, pp. 209–232, 2004.

- [133] L. D. Bui, Y. G. Kim. An obstacle-avoidance technique for autonomous underwater vehicles based on BK-products of fuzzy relation. *Fuzzy Sets and Systems*, vol. 157, no. 4, pp. 560–577, 2006.
- [134] J. Zhang, L. Q. Wang, M. J. Zhang. Path planning for underwater robot by fuzzy decision making method. *Journal* of Harbin Engineering University, vol. 22, no. 1, pp. 49–54, 2010. (in Chinese)
- [135] D. Q. Zhu, Y. Y. Yang, M. Z. Yan. Path planning algorithm for AUV based on a Fuzzy-PSO in dynamic environments. In Proceedings of the 8th IEEE International Conference on Fuzzy Systems and Knowledge Discovery, IEEE, Shanghai, China, vol. 1, pp. 525–530, 2011.
- [136] X. M. Liu, L. Peng, J. W. Li, Y. R. Xu. Obstacle avoidance using fuzzy neural networks. In Proceedings of IEEE International Symposium on Underwater Technology, IEEE, Tokyo, Japan, pp. 282–286, 1998.
- [137] H. Sayyaadi, T. Ura, T. Fujii. Collision avoidance controller for AUV systems using stochastic real value reinforcement learning method. In *Proceedings of the 39th IEEE International Conference SICE*, IEEE, Iizuka, Japan, pp. 165–170, 2000.
- [138] A. C. Schultz. Using a Genetic Algorithm to Learn Strategies for Collision Avoidance and Local Navigation, Washington, USA: Naval Research Lab Washington DC, 1990.
- [139] F. G. Ding, P. Jiao, X. Q. Bian, H. J. Wang. AUV local path planning based on virtual potential field. In *Proceed*ings of *IEEE International Conference on Mechatronics and Automation*, *IEEE*, Niagara Falls, Canada, vol. 4, pp. 1711– 1716, 2005.
- [140] H. Zou, X. Q. Bian, Z. H. Chang. A real-time obstacle avoidance method for AUV using a multibeam forward looking sonar. *Robot*, vol. 29, no. 1, pp. 82–87, 2007.
- [141] E. M. Sozer, M. Stojanovic, J. G. Proakis. Underwater acoustic networks. *IEEE Journal of Oceanic Engineering*, vol. 25, no. 1, pp. 72–83, 2000.
- [142] R. C. Joshi, V. K. Thakar. Congestion control in communication networks using discrete sliding mode control. In *Proceedings of the 31st Chinese Control Conference*, IEEE, Hefei, China, pp. 5553–5557, 2012.
- [143] X. J. O. Onunga, R. W. Donaldson. A simple packet retransmission strategy for throughput and delay enhancement on power line communication channels. *IEEE Transactions on Power Delivery*, vol. 8, no. 3, pp. 818–826, 1993.
- [144] J. Korhonen, Y. Wang. Effect of packet size on loss rate and delay in wireless links. In Proceedings of IEEE Conference on Wireless Communications and Networking, IEEE, New Orleans, USA, vol. 3, pp. 1608–1613, 2005.
- [145] Y. G. Sun, L. Wang, G. M. Xie. Average consensus in networks of dynamic agents with switching topologies and multiple time-varying delays. *Systems & Control Letters*, vol. 57, no. 2, pp. 175–183, 2008.

[146] B. Jouvencel, V. Creuze, P. Baccou. A new method for multiple AUV coordination: A reactive approach. In Proceedings of the 8th IEEE International Conference on

Emerging Technologies and Factory Automation, IEEE, Antibes-Juan les Pins, France, vol. 1, pp. 51–55, 2001.

- [147] P. K. Paim, B. Jouvencel, L. Lapierre. A reactive control approach for pipeline inspection with an AUV. In *Proceedings of MTS/IEEE OCEANS*, IEEE, Washington, USA, vol. 1, pp. 201–206, 2005.
- [148] H. Kawano. Real-time obstacle avoidance for underactuated autonomous underwater vehicles in unknown vortex sea flow by the MDP approach. In Proceedings of IEEE/RSJ International Conference on Intelligent Robots and Systems, IEEE, Beijing, China, pp. 3024–3031, 2006.
- [149] W. H. Luo, J. Peng, J. Wang. An improved parameterized approach for real time optimal motion planning of AUV moving in dynamic environment. In Proceedings of the 6th IEEE Conference on Robotics, Automation and Mechatronics, IEEE, Manila, Philippines, pp. 103–108, 2013.
- [150] L. P. Ellekilde, J. W. Perram. Tool center trajectory planning for industrial robot manipulators using dynamical systems. *The International Journal of Robotics Research*, vol. 24, no. 5, pp. 385–396, 2005.
- [151] P. Ghorbanian, S. G. Nersesov, H. Ashrafiuon. Obstacle avoidance in multi-vehicle coordinated motion via stabilization of time varying sets. In *Proceedings of American Control Conference*, IEEE, San Francisco, USA, pp. 3381– 3386, 2011.
- [152] S. G. Nersesov, P. Ghorbanian, A. G. Aghdam. Stabilization of sets with application to multi-vehicle coordinated motion. Automatica, vol. 46, no. 9, pp. 1419–1427, 2010.
- [153] A. P. Aguiar, J. P. Hespanha. Trajectory-tracking and path-following of underactuated autonomous vehicles with parametric modeling uncertainty. *IEEE Transactions on Automatic Control*, vol. 52, no. 8, pp. 1362–1379, 2005.
- [154] W. Ren, N. Sorensen. Distributed coordination architecture for multi-robot formation control. *Robotics and Au*tonomous Systems, vol. 56, no. 4, pp. 324–333, 2008.
- [155] R. Skjetne, S. Moi, T. I. Fossen. Nonlinear formation control of marine craft. In *Proceedings of the 41st IEEE Conference on Decision and Control*, IEEE, Las Vegas, USA, vol. 2, pp. 1699–1704, 2002.
- [156] H. Bai, M. Arcak, J. T. Wen. Adaptive design for reference velocity recovery in motion coordination. Systems & Control Letter, vol. 57, no. 8, pp. 602–610, 2008.
- [157] M. Arcak. Passivity as a design tool for group coordination. *IEEE Transactions on Automatic Control*, vol. 52, no. 8, pp. 1380–1390, 2007.
- [158] W. Wang, J. Q. Yi, D. B. Zhao, D. T. Liu. Design of a stable sliding-mode controller for a class of second-order underactuated system. *IEE Proceedings-Control Theory and Applications*, vol. 151, no. 6, pp. 683–690, 2004.
- [159] J. Cheng, J. Q. Yi, D. B. Zhao. Design of a sliding mode controller for trajectory tracking problem of marine vessels. *IET Control Theory & Applications*, vol. 1, no. 1, pp. 233– 237, 2007.

International Journal of Automation and Computing 13(3), June 2016

D Springer

B. Das et al. / Cooperative Formation Control of Autonomous Underwater Vehicles: An Overview

- [160] R. Ghabcheloo, A. P. Aguiar, A. Pascoal, C. Silvestre, I. Kaminer, J. Hespanha. Coordinated path-following in the presence of communication losses and time delays. *SIAM Journal on Control and Optimization*, vol. 48, no. 1, pp. 234–265, 2009.
- [161] B. Subudhi, S. S. Ge. Sliding-mode-observer-based adaptive slip ratio control for electric and hybrid vehicles. *IEEE Transactions on Intelligent Transportation Systems*, vol. 13, no. 4, pp. 1617–1626, 2012.
- [162] P. Millan, L. Orihuela, I. Jurado, F. R. Rubio. Formation control of autonomous underwater vehicles subject to communication delays. *IEEE Transactions on Control Systems Technology*, vol. 22, no. 2, pp. 770–777, 2014.
- [163] D. Angeli, E. Mosca. Lyapunov-based switching supervisory control of nonlinear uncertain systems. *IEEE Transactions on Automatic Control*, vol. 47, no. 3, pp. 500–505, 2002.
- [164] Z. L. Hu, C. Ma, L. X. Zhang, A. Halme. Distributed formation control of autonomous underwater vehicles with impulsive information exchanges and disturbances under fixed and switching topologies. In *Proceedings of IEEE the 23rd International Symposium on Industrial Electronics*, IEEE, Istanbul, Turkey, pp. 99–104, 2014.
- [165] Y. C. Cao, W. Ren. Distributed coordinated tracking with reduced interaction via a variable structure approach. *IEEE Transactions on Automatic Control*, vol. 57, no. 1, pp. 33– 48, 2012.
- [166] G. Q. Xia, C. C. Pang, J. Liu. Neural-network-based adaptive observer design for autonomous underwater vehicle in shallow water. In *Proceedings of the 9th International Conference on Natural Computation*, IEEE, Shenyang, China, pp. 216–221, 2013.
- [167] N. E. Leonard, D. A. Paley, F. Lekien, R. Sepulchre, D. M. Fratantoni, R. E. Davis. Collective motion, sensor networks, and ocean sampling. *Proceedings of the IEEE*, vol. 95, no. 1, pp. 48–74, 2007.
- [168] J. A. Marshall, M. E. Broucke, B. A. Francis. Formations of vehicles in cyclic pursuit. *IEEE Transactions on Automatic Control*, vol. 49, no. 11, pp. 1963–1974, 2004.
- [169] N. Palomeras, J. C. Garcia, M. Prats, J. J. Fernández, P. J. Sanz, P. Ridao. A distributed architecture for enabling autonomous underwater Intervention Missions. In Proceedings of the 4th Annual IEEE Systems Conference, IEEE, San Diego, USA, pp. 159–164, 2010.
- [170] B. K. Sahu, B. Subudhi, B. K. Dash. Flocking control of multiple autonomous underwater vehicles. In *Proceed*ings of Annual IEEE India Conference, IEEE, Kochi, India, pp. 257–262, 2012.
- [171] S. Y. Liu, D. W. Wang, E. Poh. Non-linear output feedback tracking control for AUVs in shallow wave disturbance condition. *International Journal of Control*, vol. 81, no. 11, pp. 1806–1823, 2008.
- [172] S. Y. Liu, D. W. Wang, E. K. Poh, Y. G. Wang. Dynamic positioning of AUVs in shallow water environment: Observer and controller design. In Proceedings of IEEE/ASME International Conference on Advanced Intelligent Mechatronics, IEEE, Monterey, USA, pp. 705–710, 2005.

- [173] A. Pant, P. Seiler, K. Hedrick. Mesh stability of look-ahead interconnected systems. *IEEE Transactions on Automatic Control*, vol. 47, no. 2, pp. 403–407, 2002.
- [174] P. M. Lee, S. W. Hong, Y. K. Lim, C. M. Lee, B. H. Jeon, J. W. Park. Discrete-time quasi-sliding mode control of an autonomous underwater vehicle. *IEEE Journal of Oceanic Engineering*, vol. 24, no. 3, pp. 388–395, 1999.
- [175] S. Negahdaripour, S. Cho, K. Y. Joon. Controller design for an autonomous underwater vehicle using nonlinear observers. International Journal of Ocean System Engineering, vol. 1, pp. 16–27, 2011.
- [176] J. C. Kinsey, Q. J. Yang, J. C. Howland. Nonlinear dynamic model-based state estimators for underwater navigation of remotely operated vehicles. *IEEE Transactions on Control Systems Technology*, vol. 22, no. 5, pp. 1845–1854, 2014.
- [177] J. B. de Sousa, F. L. Pereira. On coordinated control strategies for networked dynamic control systems-An application to AUVs. In Proceedings of International Symposium of Mathematical Theory of Networks and Systems, Pereslavl-Zalesskii, Russia, 2002.
- [178] E. Burian, D. Yoerger, A. Bradley, H. Singh. Gradient search with autonomous underwater vehicles using scalar measurements. In Proceedings of IEEE Symposium on Autonomous Underwater Vehicle Technology, IEEE, Monterey, USA, pp. 86–98, 1996.
- [179] A. Speranzon. On Control under Communication Constraints in Autonomous Multi-robot Systems, Ph. D. dissertation, Licentiate Thesis, Stockholm, Sweden, 2004.
- [180] B. Garau, M. Bonet, A. Alvarez, S. Ruiz, A. Pascual. Path planning for an AUV in realistic oceanic current fields: Application to glider in the western Mediterranean sea. *Journal of Maritime Research*, vol. 6, no. 2, pp. 5–22, 2009.
- [181] D. P. Horner, A. J. Healey, S. P. Kragelund. AUV experiments in obstacle avoidance. In *Proceedings of IEEE OCEANS*, IEEE, Washington, USA, vol. 2, pp. 1464–1470, 2005.
- [182] H. Bing, L. Gang, G. Jiang, W. Hong, N. Nan, L. Yan. A route planning method based on improved artificial potential field algorithm. In Proceedings of IEEE the 3rd International Conference on Communication Software and Networks, IEEE, Xi'an, China, pp. 550–554, 2011.
- [183] A. Speranzon. Coordination, Consensus and Communication in Multi-Robot Control Systems, Ph. D. dissertation, Licentiate Thesis, Stockholm, Sweden, 2006.
- [184] X. Z. Cheng, H. N. Shu, Q. L. Liang, D. H. C. Du. Silent positioning in underwater acoustic sensor networks. *IEEE Transactions on Vehicular Technology*, vol. 57, no. 3, pp. 1756–1766, 2008.
- [185] A. P. Aguiar, A. M. Pascoal. Dynamic positioning and waypoint tracking of under actuated AUVs in the presence of ocean currents. In *Proceedings of the 41st IEEE Conference* on *Decision and Control*, IEEE, Las Vegas, USA, vol. 2, pp. 2105–2110, 2002.

🖉 Springer

- [186] B. Subudhi, K. Mukherjee, S. Ghosh. A static output feedback control design for path following of autonomous underwater vehicle in vertical plane. *Ocean Engineering*, vol. 63, pp. 72–76, 2013.
- [187] X. J. Jing, C. Y. Wang, D. L.Tan. Artificial coordinating field based real-time coordinating collision-avoidance planning for multiple mobile robots. *Control Theory and Applications*, vol. 21, no. 5, pp. 757–764, 2004. (in Chinese)
- [188] D. P. Horner, A. J. Healey. Use of artificial potential fields for UAV guidance and optimization of WLAN communications. In *Proceedings of IEEE/OES Autonomous Underwater Vehicles*, IEEE, Sebasco, USA, pp. 88–95, 2004.
- [189] H. L. Xu, Y. P. Li. An immune genetic algorithm for AUV local path planning. In Proceedings of the 20th International Offshore and Polar Engineering Conference, ISOPE, Beijing, China, pp. 396–400, 2010.
- [190] D. R. Blidberg. The development of autonomous underwater vehicles (AUV); a brief summary. In Proceedings of IEEE International Conference on Robotics and Automation, IEEE, Lee New Hampshire, USA, 2001.
- [191] C. C. Sotzing, D. M. Lane. Improving the coordination efficiency of limited-communication multi-autonomus underwater vehicle operations using a multiagent architecture. *Journal of Field Robotics*, vol. 27, no. 4, pp. 412–419, 2010.
- [192] F. Fahimi. Sliding mode formation control for underactuated surface vessels. *IEEE Transactions on Robotics*, vol. 23, no. 3, pp. 617–622, 2007.
- [193] J. P. Desai, J. P. Ostrowski, V. Kumar. Modeling and control of formations of nonholonomic mobile robots. *IEEE Transactions on Robotics and Automation*, vol. 17, no. 6, pp. 905–908, 2001.
- [194] Y. Y. Dai, S. G. Lee. The leader-follower formation control of nonholonomic mobile robots. International Journal of Control, Automation, and Systems, vol. 10, no. 2, pp. 350– 361, 2012.
- [195] L. Sorbi, G. P. De Capua, L. Toni, J. G. Fontaine. Target detection and recognition: A mission planner for Autonomous Underwater Vehicles. In *Proceedings of IEEE OCEANS*, IEEE, Waikoloa, USA, pp. 1–5, 2011.
- [196] R. Ghabcheloo, I. Kaminer, A. P. Aguiar, A. Pascoal. A general framework for multiple vehicle time-coordinated path following control. In Proceedings of the 2009 conference on American Control Conference, IEEE, Piscataway, USA, pp. 3071–3076, 2009.
- [197] W. Yan, W. S. Yan, Y. T. Wang, Y. Yao. Formation control of multiple underactuated autonomous underwater vehicles with communication time delays. *Fire Control & Command Control*, vol. 36, no. 6, pp. 52–55, 2011. (in Chinese)
- [198] D. B. Gu, H. S. Hu. Using fuzzy logic to design separation function in flocking algorithms. *IEEE Transactions on Fuzzy Systems*, vol. 16, no. 4, pp. 826–838, 2008.
- [199] R. O. Saber. Flocking for multi-agent dynamic systems: Algorithms and theory. *IEEE Transactions on Automatic Control*, vol. 51, no. 3, pp. 401–420, 2006.

- International Journal of Automation and Computing 13(3), June 2016
 - [200] D. B. Gu, Z. Y. Wang. Leader-follower flocking: Algorithms and experiments. *IEEE Transactions on Control Systems Technology*, vol. 17, no. 5, pp. 1211–1219, 2009.
 - [201] P. Rattanasiri, P. A. Wilson, A. B. Phillips. Numerical investigation of a fleet of towed AUVs. Ocean Engineering, vol. 80, pp. 25–35, 2014.
 - [202] B. K. Sahu, M. M. Gupta, B. Subudhi. Fuzzy separation potential function based flocking control of multiple AUVs. In Proceedings of IEEE IFSA World Congress and NAFIPS Annual Meeting, IEEE, Edmonton, Canada, pp. 1429–1434, 2013.
 - [203] X. B. Xiang, G. H. Xu, Q. Zhang, Z. H. Xiao, X. H. Huang. Coordinated control for multi-AUV systems based on hybrid automata. In Proceedings of IEEE International Conference on Robotics and Biomimetics, IEEE, Sanya, China, pp. 2121–2126, 2007.
 - [204] S. Sariel, T. Balch, N. Erdogan. Naval mine countermeasure missions. *IEEE Robotics & Automation Magazine*, vol. 15, no. 1 pp. 45–52, 2008.
 - [205] F. Y. Bi, Y. J. Wei, J. Z. Zhang, W. Cao. Position-tracking control of underactuated autonomous underwater vehicles in the presence of unknown ocean currents. *IET Control Theory & Applications*, vol. 4, no. 11, pp. 2369–2380, 2010.
 - [206] D. H. Wang, J. Yuan, J. Xu, Z. H. Zhou. Finite-time formation control for autonomous underwater vehicles based on hierarchical leader-follower. *Applied Mechanics and Materials*, vol. 541–542, pp. 1164–1167, 2014.
 - [207] L. Ding, Q. L. Han, G. Guo. Network-based leaderfollowing consensus for distributed multi-agent systems. *Automatica*, vol. 49, no. 7, pp. 2281–2286, 2013.
 - [208] P. W. Kimball, S. M. Rock. Mapping of translating, rotating icebergs with an autonomous underwater vehicle. *IEEE Journal of Oceanic Engineering*, vol. 40, no. 1, pp. 196–208, 2015.
 - [209] T. Matsuda, T. Maki, T. Sakamaki, T. Ura. State estimation and compression method for the navigation of multiple autonomous underwater vehicles with limited communication traffic. *IEEE Journal of Oceanic Engineering*, vol. 40, no. 2, pp. 337–348, 2015.
 - [210] K. Shojaei, M. M. Arefi. On the neuro-adaptive feedback linearising control of underactuated autonomous underwater vehicles in three-dimensional space. *IET Control Theory* & Applications, vol. 9, no. 8, pp. 1264–1273, 2015.
 - [211] S. H. Li, X. Y. Wang, L. J. Zhang. Finite-time output feedback tracking control for autonomous underwater vehicles. *IEEE Journal of Oceanic Engineering*, vol. 40, no. 3, pp. 727–751, 2015.
 - [212] Y. Q. Xia, Y. L. Gao, L. P. Yan, M. Y. Fu. Recent progress in networked control systems-A survey. *International Jour*nal of Automation and Computing, vol. 12, no. 4, pp. 343– 367, 2015.
 - [213] B. Das, B. Subudhi, B. B. Pati. Adaptive sliding mode formation control of multiple underwater robots. Archives of Control Sciences, vol. 24, no. 4, pp. 515–543, 2014.

224

🖄 Springer

- [214] B. Das, B. Subudhi, B. B. Pati. Co-operative control coordination of a team of underwater vehicles with communication constraints. *Transactions of the Institute of Measurement and Control*, vol. 38, no. 4, pp. 463–481, 2016.
- [215] D. H. Shin, S. T. Kwon, S. H. Park, M. G. Joo. Fuzzy state feedback control for way-point tracking of autonomous underwater vehicle. *International Journal of Control and Au*tomation, vol. 6, no. 1, pp. 119–130, 2013.
- [216] B. Das, B. Subudhi, B. B. Pati. Employing nonlinear observer for formation control of AUVs under communication constraints. *International Journal of Intelligent Unmanned Systems*, vol. 3, no. 2–3, pp. 122–155, 2015.



Bikramaditya Das received the B. Tech. degree in electronics and telecommunications engineering from the Biju Patnaik University of Technology, Rourkela, M. Tech. degree in wireless communication from Department of Electrical Engineering, National Institute of Technology Rourkela, India in 2010, and the Ph.D. degree in control under communication constraints

from Department of Electrical Engineering, Veer Surendra Sai University of Technology Odisha, Burla in 2016. He is currently working as assistant professor at Department of Electronics and Telecommunication Engineering, Veer Surendra Sai University of Technology Odisha, Burla. He is an associate member of IEI.

His research interests include formation control of autonomous underwater vehicles, robotics and wireless communication.

E-mail: adibik09@gmail.com ORCID ID: 0000-0001-9734-806X



Bidyadhar Subudhi received the B. Eng. degree in electrical engineering from National Institute of Technology (NIT), Regional Engineering College (REC), India in 1988, the M. Eng. degree in control and instrumentation from Indian Institute of Technology Delhi (IIT), India in 1993, and the Ph. D. degree in control system engineering from University of Sheffield, UK in 2003. He was a post-doctoral research fellow at Department of Electrical & Computer Engineering, National University of Singapore (NUS), Singapore from May to November in 2005. He was a visiting professor in University of Saskatchewan, Canada from May to June in 2009 and also at Asian Institute of Technology (AIT) Bangkok from January to May in 2013. He is a professor at Department Electrical Engineering, National Institute of Technology (NIT), India. He is currently coordinator of Centre of Excellence, Renewable Energy System, Centre of Excellence, Industrial Electronics & Robotics, and head of Computer Centre at NIT. He is also the principal investigator in several research projects funded by Defense Research Development Organization (DRDO), Council of Scientific and Industrial Research (CSIR) and Department of Science and Technology (DST), including an international cooperation project under the UK India Education Research Initiative (UKIERI) scheme. He has published 40 journal papers in prestigious journals such as IEEE Transactions, IET and Elsevier and presented 70 research papers in many international conferences both in India and abroad. He has edited one book and contributed 3 book chapters. He chaired a number of technical sessions in international conferences. He is a fellow of the IET (UK) and senior member of IEEE (USA). He is a regular reviewer of IEEE Transaction on Systems, Man, & Cybernetics, Power Delivery, Control System Technology and Neural Networks and Automatic Control.

His research interests include system identification and adaptive control, networked control system, control of flexible and under water robots, control of renewable energy systems, estimation and filtering with application to power system.

E-mail: bidyadhar@nitrkl.ac.in (Corresponding author) ORCID ID: 0000-003-4383-6783



Bibhuti Bhusan Pati is currently working as a professor in Electrical Engineering Department, Veer Surendra Sai University of Technology Odisha, Burla. He is the fellow of Institution of Engineers, member Indian Society for Technical Education, Bigyan Academy, and Engineering Congress. He has published more than 100 papers in reputed journals and conferences.

He is the investigator of many AICTE sponsored projects. His research of interests include control system and applications to power system.

E-mail: pati_bibhuti@rediffmail.com